Integrated reports, external assurance and financial performance

An empirical analysis on North American firms

Orhan Akisik and Graham Gal Department of Accounting, University of Massachusetts, Amherst, Massachusetts, USA Empirical analysis on North American firms

317

Received 27 February 2019 Revised 18 June 2019 Accepted 17 July 2019

Abstract

Purpose – The purpose of this paper is to examine the relationship between integrated reports, external assurance and financial performance for North American firms between 2011 and 2016.

Design/methodology/approach – Corporate websites were examined for disclosures which included both financial and non-financial information. Compustat North America and Global Reporting Initiative (GRI) websites provided additional data for the analysis.

Findings – Using a panel data analysis, the results provide evidence that there is a significant positive association between integrated reports and multiple measures of financial performance. Moreover, this positive effect is enhanced when integrated reports are assured by accounting firms.

Research limitations/implications – There are relatively a small number of firms that do this kind of reporting. A major limitation of the study is the small sample size.

Practical implications – As stakeholders find information in integrated reports relevant, there needs to be standardization on their content and level of assurance. Standard setters and regulators should be involved in setting these standards and assurance guidelines.

Social implications – Although it is clear that there is a cost to firms which produce integrated reports, the benefits to society may outweigh these costs. This may go beyond the benefits to shareholders as they make investment decisions.

Originality/value – According to the knowledge of the authors, this is the first study that examines the impact of integrated reports and external assurance on financial performance for North American firms.

Keywords Corporate social responsibility, Financial performance, External assurance, Integrated report

Paper type Research paper

1. Introduction

There is ample evidence that investors view information in financial statements as valuerelevant (Francis and Schipper, 1999). Studies provide growing evidence that investors also consider non-financial information useful in their decisions (Barth and McNichols, 1994; Amir and Lev, 1996; Elliott *et al.*, 2014). One area that continues to be discussed as being value-relevant is the degree to which firms engage in socially responsible actions (Griffin and Mahon, 1997). As an example, Griffin *et al.* (2012) provide evidence that shareholders

The authors thank the editors, two anonymous reviewers, Jeremy Bentley and Suleyman Ozmucur for insightful comments on a previous version of this paper.



Sustainability Accounting, Management and Policy Journal Vol. 11 No. 2, 2020 pp. 317-350 © Emerald Publishing Limited 2040-8021 DOI 10.1108/SAMPJ-02-2019-0072



SAMPJ 11,2

318

consider information on greenhouse gas emissions in their investment decisions. However, the debate continues on what actions are socially responsible (Dahlsrud, 2008).

Actions included under the broad heading of corporate social responsibility (CSR) can include some behaviors that may be closer to image management than to actually being socially responsible (Bebbington et al., 2008; Gal, 2018; Moir, 2001). Petrenko et al. (2016) find evidence of firms disclosing corporate social responsible actions to enhance their image without focusing on financial performance. In support of this view, there are studies indicating that CSR can be used as a marketing strategy (Polonsky, 1995). These CSR disclosures may make firms appear to be more legitimate (Herzig and Schaltegger, 2006), but may not improve their financial performance (Akisik and Gal, 2017; Dhaliwal et al., 2011; McWilliams and Siegel, 2001). This has provided support for arguments that question whether management should really be engaged in actions which may not have a positive impact on financial performance, as their only goal is to maximize shareholder wealth (Friedman, 1970). The fact that there is no consensus on the impact of CSR on financial performance could be owing to two issues. First, in contrast to financial statements, nonfinancial disclosures are voluntary (Abeysekera, 2008). Second, there is not a single accepted view about actions which are socially responsible (Dahlsrud, 2008; Whitehouse, 2006). This may make it difficult for investors to determine the impact that a firm's CSR actions will have on financial performance.

Providing investors (or any stakeholders) with better or more relevant information can certainly reduce asymmetry, uncertainty and therefore risk (ACCA, 2017a; Akins, 2018; Cuadrado-Ballesteros *et al.*, 2017; García-Sánchez and Noguera-Gámez, 2017b)[1]. In certain industries, CSR actions may have a particularly significant impact on a firm's future performance. For instance, financial performance of firms in the petroleum industry may be significantly impacted by actions that could be included in a CSR disclosure (Schneider *et al.*, 2011). Thus, both CSR disclosures and financial statements contain information that can be relevant for investors, as they provide different but potentially complementary views of firms' prospects (Flammer, 2013). Therefore, there is a growing call for a single report which integrates these different perspectives (The International Integrated Reporting Council, 2013).

To understand whether investors consider integrated reports (IRs) to be value-relevant, this study examines the relationship between North American firms' IRs and financial performance for the period between 2011 and 2016. The International Integrated Reporting Council (IIRC) has supported the creation of a single report which would combine both financial and non-financial measures (The International Integrated Reporting Council, 2013). However, IRs have a similar issue with CSR reports, in that there is no agreed-upon framework or template for such a comprehensive disclosure (Abeysekera, 2013; McNally *et al.*, 2017; de Villers *et al.*, 2017). To improve the quality of IRs, there are some suggestions that information contained in them meets tests such as materiality, reliability, completeness and consistency (ACCA, 2018). With a central focus of an IR on a firm's value creation activities, there is an argument that these reports should be relevant to investors (ACCA, 2018). While financial statements will have independent assurance, the non-financial information in IRs may not. Therefore, assurance on CSR information and other non-financial information to be included in an IR may increases the use of and credibility of these disclosures by investors.

Certainly, a review of any non-financial disclosure in general, and CSR disclosures in particular, is not required, and so in many cases they are not reviewed. The Global Reporting Initiative (GRI) has called for some assurance on disclosures which use their format (Global Reporting Initiative, 2013). It would seem that a review of any non-financial disclosure would add to its credibility (Holm and Birkholm-Laursen, 2007; Manetti and Becatti, 2009; Martínez-Ferrero and García-Sánchez, 2018). However, there is a question of



how this review would be performed and who should provide this review (Aguinis and Glavas, 2012; Cohen and Simnett, 2015; Simnett and Huggins, 2015). As CSR disclosures have become more prevalent, auditors have added assurance of CSR reports to their practice (O'Dwyer and Owen, 2005). O'Dwyer *et al.* (2011) argue that when auditors add assurance for CSR reports to their practice, this adds to the legitimacy of firms as general assurance providers and also to the legitimacy of this type of assurance. Therefore, the assurance may provide users with a basis for accepting the firms' CSR disclosures without looking at the detail (Akisik and Gal, 2014). This is similar to findings which suggest that users of financial statements focus on the review about financial performance (Asare and Wright, 2012; Schaub, 2006).

This study is the first to examine the relationships between IRs, external assurance and financial performance. It makes four major contributions to the literature. First, as a relatively new concept, there has not been a great deal of research on IRs (Maniora, 2017). Prior studies generally focus on theoretical issues and survey about IRs (Eccles and Serafeim, 2011; The International Integrated Reporting Council, 2013; KPMG, 2012; Maniora, 2017; Serafeim, 2015). Are IRs really important for financial performance? The goals of this study is to shed light on this issue. Second, the study extends previous research on IRs to the role of external assurance and investigates its impact on the relationship between IRs and financial performance (Dando and Swift, 2003; Eccles et al., 2019; Wallage, 2000). As a third contribution, the study examines whether the credibility and the reliability of CSR reports provided by accounting firms' assurance extends to IRs (Cho et al., 2014). There has been a long-standing debate about the impact of CSR actions on financial performance for firms in certain sectors (Kim and Venkatachalam, 2011). There is evidence that the business operations of firms in pollution-prone sectors are closely screened by stakeholders (Magness, 2010; Thompson and Cowton, 2004). Therefore, many firms operating in these sectors inform stakeholders about their efforts to reduce the negative impacts of their operations on environment and society. The study's fourth contribution is to analyze the relationships between IRs, external assurance and financial performance in pollution-prone sectors.

Data for this study come from various sources. The financial data are from Compustat North America, and the non-financial data about CSR reviews come from the GRI website. In addition, data on IRs and their assurance are hand-collected. Using a panel data fixed effects model for the period between 2011 and 2016, the results show that IRs are positively and significantly related to financial performance, and that this positive relationship is enhanced by external assurance provided by accounting firms. Furthermore, when the analysis is replicated for pollution-prone sectors similar results are obtained.

The remainder of the paper is organized as follows. Section 2 examines literature on stakeholder theory as it relates to IRs and the impact of assurance on these reports. Section 3 develops hypotheses. Section 4 presents the methodology and econometric model. Section 5 discusses the results of empirical analyses, including robustness tests and additional analyses. Finally, Section 6 presents conclusions and implications.

2. Literature review

2.1 Integrated reports and stakeholders

Arguments are made that a firm's only responsibility is to provide relevant information to shareholders as they provide the capital necessary to run the firm (Friedman, 1970). Some studies have concluded that altering the firm's business practice to become more socially responsible will diminish financial performance (Griffin and Mahon, 1997). While this argument may suggest that firms only need to disclose financial information, as this provides value-relevant information for investors, there are at least two arguments to suggest that additional disclosures are appropriate. First, there is evidence that shareholders



Empirical analysis on North American firms

SAMPJ 11,2

320

find additional information to be value-relevant; this will impact their investment decisions (Francis and Schipper, 1999). A second argument, and just as compelling, is that other groups of stakeholders have a legitimate stake in the firm's future, and therefore, they have an interest in disclosures that are relevant to their position in the firm. One type of additional information stakeholders consider relevant to their decisions falls under the broad heading of social responsibility disclosures

Shareholders have an interest in the future of the firm and the risk to their invested capital (Gamble and Kelly, 2001). While information in financial statements can help investors evaluate certain types of risk, information contained in CSR reports may also be useful for determining credit risk and therefore returns to shareholders (García-Sánchez and Noguera-Gámez, 2017a). For instance, Heyes (1996) finds that firms with higher potential for environmental damage can be charged higher interest rates. In a later study, Thompson and Cowton (2004) determine that lenders increase the scrutiny of firms with higher environmental risk. Thus, disclosures which indicate control over potential environmental risk may prove beneficial to shareholders, as these actions can reduce certain costs of doing business (Cormier *et al.*, 1993). Therefore, boards of directors are increasingly making additional disclosures about the firm's social responsible activities (Frias-Aceituno *et al.*, 2013). It is evident that shareholders may also find this information useful.

Despite the arguments of Friedman (1970), there are views that firms should consider the needs or concerns of a broader group of stakeholders. The relationship between a firm and the needs of its stakeholders can be quite complex, as there are many different types of stakeholders and therefore different ways to engage them (Friedman and Miles, 2002; Scholtens and Zhou, 2008). Furthermore, supporting the needs of one group of stakeholders may be detrimental to another (Köbberling and Wakker, 2004). For example, an attempt to alter a firm's operations to meet the needs of any particular group of stakeholders could reduce their ability to provide long-term employment and meet the needs of employees. However, Gibson (2000) argues that there is a moral basis for firms to consider more than just shareholders in their operational decisions. Stakeholder theory suggests that there are ways to measure performance other than financially and to support many views of the firm's role in society (Freeman, 1984; Friedman and Miles, 2002; Kaler, 2006; Polonsky, 1995). To evaluate the impact of a firm's actions directed at different stakeholder groups, it is important to match specific socially responsible actions to specific groups of stakeholders when looking for performance measures (Wood and Jones, 1995). For instance, providing products that meet the needs of customers may produce loyalty and therefore an increase in sales for those products (Brown and Dacin, 1997; Madden et al., 2012; Öberseder et al., 2013; Romani et al., 2013). This increased loyalty may also provide sustainable financial results. Campbell (1997) suggests that a shareholder view of performance will result in a short-term view of performance, and that a broader stakeholder perspective will allow a firm to survive and be successful in the long term. This is similar to the view of customers as stakeholders who provide long-term support for a firm that provides products which meet their needs. Stakeholder theory argues that firms, which consider more than just shareholders in their operations, will receive support from a variety of stakeholder groups (Freeman, 1984; Friedman and Miles, 2002). In addition to considering these multiple stakeholders, disclosure of information relevant to them is critical.

For socially responsible firms to receive support from stakeholders, there is a need to make them aware of the firms' CSR activities. Herzig and Schaltegger (2006) argue that firms can appear to be a more legitimate member of society by disclosing their non-financial results. As these disclosures are voluntary and concern activities that are generally not



legally required, this signals a desire to communicate with a broader group of stakeholders and an attempt to be viewed as a legitimate member of society (Benn and Bolton, 2011; Carroll and Shabana, 2010; Hockerts and Moir, 2004). Thus the management's decision to adopt a stakeholder view of the firm's goals and to provide additional disclosures can be seen as an attempt to change the society's view of the firm.

A concern in earlier discussion is that stakeholders have difficulty in relating CSR information to a variety of performance measures. While this concern may be particularly relevant for shareholders, other stakeholders may be interested in the long-term viability of the firm and how CSR actions may impact its future (Cahan *et al.*, 2016; Martínez-Ferrero and García-Sánchez, 2017). For this reason (and potentially others), there has been a growing call for management disclosure which would integrate both financial and non-financial information (Eccles and Krzus, 2010, 2014; Eccles *et al.*, 2015).

An IR is a corporate report which combines the multiple, "[...] factors that materially affect the ability of an organization to create value over time" (The International Integrated Reporting Council, 2013, p. 2). Early in the process of advocating for an IR, The International Integrated Reporting Council (2003) argues that this disclosure would be a concise communication about how a firm's strategy, governance, performance and prospects, in the context of its external environment, lead to the creation of value over the short, medium and long term. Because the IRs would include information from multiple sources, there are some studies that consider it a complement to these sources (ACCA, 2013; Remgro, 2017). The creation of a single report, which combines alternative sources, allows boards to make decisions about which disclosures are appropriate (Abeysekera, 2008; Frías-Aceituno *et al.*, 2013)[2]. This raises an issue about the format of an IR, which is similar to the issue that was considered when disclosure of CSR information was just beginning[3]. Therefore, there are discussions about creating a format for IR (Abeysekera, 2013),

Advocates of IRs indicate that these single reports will focus on value creation and therefore the future success of the firm (ACCA, 2018). The focus on IRs is on combining multiple sources of information on the firm's value and strategies. Thus an IR improves communication from the management and would be particularly relevant to investors (ACCA, 2017b; Cuadrado-Ballesteros *et al.*, 2017; García-Sánchez and Noguera-Gámez, 2017b). Although there are perceived benefits to integrating both financial and non-financial information in a single report, IRs have also been subject to some of the same criticisms that were leveled at CSR reports (Cuadrado-Ballesteros *et al.*, 2017).

Because IRs are still in their infancy, there is a need for research both on its format and for guidance on their preparation (Adams and Narayanan, 2007; de Villers et al., 2017; Martínez-Ferrero et al., 2018). The lack of guidance on the content of IRs makes them difficult to prepare (Chaidali and Jones, 2017; Maniora, 2017; Velte and Stawinoga, 2017). As IRs do not have an agreed-upon format, early adopters may be seen as being more legitimate and socially responsible than late adopters (Arguelles et al., 2017; Frías-Aceituno et al., 2014). This is similar to the benefit attributed to firms with a long history of CSR disclosures as the willingness to provide socially responsible information improves a firm's perceived integrity (Vanhamme and Grobben, 2009). While producing IRs may be difficult for firms, there is some reason to believe that investors will find these reports value-relevant. Investors may benefit from IRs because they can reduce information asymmetry about the impact of nonfinancial information on financial performance (Cosma et al., 2018; Cuadrado-Ballesteros et al., 2017; Frías-Aceituno et al., 2013; Martínez-Ferrero and García-Sánchez, 2017; Schadewitz and Niskala, 2010). However, because the content of IRs makes them attractive to investors, the assurance of the information contained becomes critical (ACCA, 2017b, Eccles et al., 2019; Moroney et al., 2012).



Empirical analysis on North American firms

SAMPJ 11,2

322

2.2 Assurance of CSR disclosures and integrated reports

Firms have been required to disclose financial information for some time (United States Congress, 1934). In contrast to social responsibility and sustainability disclosures, financial statements have a very specific format and set of procedures to provide assurance. Any firm, which wants to maintain a positive image with its broad group of stakeholders, must provide some form of sustainability disclosure (EY, 2014). Because disclosures of socially responsible actions can be important for different stakeholders, there has been a call to include some level of assurance (Monaghan, 2004). The GRI has called for some type of assurance (preferably a third-party) of CSR disclosures (Global Reporting Initiative, 2013). Cuadrado-Ballesteros *et al.* (2017), Hodge *et al.* (2009), Holm and Birkholm-Laursen (2007) and Simnett *et al.* (2009a, 2009b) confirm that some level of assurance provided by an auditor, industry specialist or engineering firm establishes the credibility of the CSR disclosure (Akisik and Gal, 2014; Martínez-Ferrero and García-Sánchez, 2018; Martínez-Ferrero *et al.*, 2018). Assurance of CSR disclosures may also add to stakeholders' understanding of it because of the nature of the data contained in these reports.

The information in CSR disclosures is varied because it is compiled for a broad group of stakeholders with diverse interests and so can be difficult to interpret (Adams and Narayanan, 2007). Therefore, assurance on their CSR disclosure is critical for stakeholders for interpreting their complex and technical content (Morsing et al., 2008; Schneider et al., 2011). CSR disclosures are not mandatory and so their assurance is also not required. However, researchers argue that users would place more reliance on these reports if they were assured (Casey and Grenier, 2015; Bagnoli and Watts, 2016). Similarly, the assurance of IRs could also make them more accepted (Cheng et al., 2014) as a credible source of information (Baron, 2014; Cheng et al., 2015; Hodge et al., 2009; McCuaig, 2010; Serafeim, 2015). Investors, who are viewed as the primary users of IRs, may be particularly interested in assurance of these reports (ACCA, 2017b). However, any independent assurance of IRs would increase fees corporations already pay for assurance services, so boards must see assurance as providing benefits to shareholders despite potential for this cost to be seen as a negative value (ACCA, 2018; García-Sánchez et al., 2019)[4]. The additional costs could create a tension between certain stakeholders who would value an IR which combines many indicators of the potential for future success and shareholders who would be reluctant to pay for assurance of IRs. This would be particularly relevant when firms may have already hired an independent party to provide assurance for financial statements and CSR reports. The development of an IR does include information from these two sources, and therefore, a significant portion of the IR already has some level of assurance. It may be expected that firms that have their CSR reports assured by third-party assurance providers are likely to issue IRs (Sierra-García et al., 2015). In addition, there is evidence that users view more complex IRs as more reliable (du Toit, 2017), but this could make them less readable. While there is no description of the metrics to be used, there has been a call for firms to provide some level of assurance for their IR (Briem and Wald, 2018; Dumay and Dai, 2017). This does raise a potential problem as there is evidence that owing to the complexity of financial statements, investors are more focused on their assurance than on their content (Schaub, 2006). Researchers have also demonstrated that assurance on stand-alone CSR reports has a significant impact on financial performance (Akisik and Gal, 2014, 2017). Therefore, while IRs should provide value-relevant information to investors, their assurance can be expected to provide additional information.

3. Hypotheses development

3.1 Effect of integrated reports on financial performance

The studies about the relationship between CSR performance and stock market effects go back to 1970s (Alexander and Buchholz, 1978; Moskowitz, 1972; Vance, 1975). Prior studies



about the relationship between market performance of common stocks and CSR provide contradictory results. Using UK and US data, Cox et al. (2004) document that a firm's poor CSR performance leads to a reduction in the number of long-term institutional investors holding the stock (Brammer et al., 2006). Coram et al. (2009) find that non-financial performance indicators significantly affect stock prices. In addition, they provide evidence that an assurance report on disclosure of voluntarily non-financial performance indicators significantly influence stock price estimates. However, this is only for positive non-financial performance indicators, suggesting that the value of assurance is context-specific. In a study for highly ranked (very large) US firms, Amato and Amato (2012) find a positive effect of corporate environmental policy on stock prices. They argue that the positive effect on stock prices could arise from socially conscious investors rewarding these firms by purchasing their stocks or stock purchases by investors who expect positive consumers' responses to firms' products. Elliott et al. (2014) argue that many investors regularly evaluate firms' CSR performance indicators along with traditional financial performance measures when making investment decisions. Grewal et al. (2017) find that firms voluntarily disclosing more information identified by the Sustainability Accounting Standards Board have higher stock price informativeness. For Canadian firms listed on the Toronto Exchange, Berthelot et al. (2012) provide evidence that the disclosure of sustainability reports has a positive impact on investors, indicating that the firms creating CSR reports receive a significant premium in the market. In contrast to this, findings of a study by Moneya and Cuellar (2009) suggest that investors attach a positive value to financial environmental disclosures (those that have an impact on financial performance) rather than environmental disclosures that do not have a direct impact on financial performance. Because CSR information has been found to impact financial performance, there has been a suggestion that an IR which combines both firm's financial information and non-financial information would be valuable to investors.

This leads to the view that IRs can create value-relevant information over the short, medium and long term (The International Integrated Reporting Council, 2013). While it is argued that IRs primarily address investors, all stakeholders might benefit from them (Eccles and Krzus, 2014; Velte and Stawinoga, 2017). The few studies on IRs and financial performance indicate that they are positively associated. For example, Churet *et al.* (2014) find no conclusive evidence that IRs are associated with firms achieving higher performance measured by return on invested capital. However, when analyzing results by sector, they find a positive association between IRs and financial performance in the health-care and information technology sectors. Knauer and Serafeim (2014) provide evidence that firms that engage in integrated thinking and reporting attract long-term investors[5]. Serafeim (2015) documents a positive association between IRs and investor clientele, suggesting that firms' IRs are value-relevant. A number of proponents of IRs argue that IRs are a more effective way of communicating firms' capabilities (The International Integrated Reporting Council, 2003). This could lead to an increase in performance (Serafeim, 2015). In light of the views above, the following hypothesis are examined:

H1. Ceteris paribus, IRs are positively associated with financial performance.

3.2 Role of external assurance on the relationship between integrated reports and financial performance

Voluntary disclosures are made and an external assurance on them is adopted only if their benefits outweigh their costs (Demartini and Trucco, 2017). Even though external assurance might enhance the credibility of CSR reports, it leads to a significant additional cost over and above audit fees, which could negatively affect shareholder value (ACCA, 2018)[6].



Empirical analysis on North American firms

SAMPJ 11,2 However, some authors (De and Sen, 2002; Firth and Smith, 1992) are of the opinion that the cost can be recovered as a result of enhanced credibility because assured reports will have a positive signaling effect to the market (García-Sánchez and Noguera-Gámez, 2017a; Mock *et al.*, 2007).

Besides its cost, the quality of external assurance is also an important factor taken into account by client-firms and stakeholders (Cohen and Simnett, 2015; de Villiers *et al.*, 2014). Is there any quality difference of external assurance in terms of the type of assurance providers? (Ferguson and Pundrich, 2015; Perego, 2009) It is argued that accounting firms, in particular Big-4, have some advantage with respect to the quality of assurance services over other assurance providers (engineering and small consultancy-boutique firms). This is because they provide their services in accordance with well-developed global standards, comply with independence and ethical requirements and have mechanisms to ensure the quality of the engagements (Cuadrado-Ballesteros *et al.*, 2017; Martínez-Ferrero *et al.*, 2018; Simnett *et al.*, 2009b). According to Perego and Kolk (2012), these differences are likely to result from conservative and cautious approaches employed by accounting firms when they assess the extent of assurance that can be provided in the face of disparate and voluntary reporting standards that firms follow when issuing CSR reports (Peters and Romi, 2015).

In contrast to these views, some authors contend that when it comes to "recommendations and opinions" contained in a statement, the assurance services provided by non-accounting firms might be more elaborate and informative, as they possess higher-level expertise of the subject matter (Deegan *et al.*, 2006; Hodge *et al.*, 2009; Mock *et al.*, 2013; Perego, 2009; Simnett *et al.*, 2009a). In other words, in contrast to accounting firms, consultancy firms are believed to focus more on completeness, fairness and overall balance in the opinion statements, which could be favorably regarded by users of these reports[7].

Neither IRs nor their external assurances are required by law in the USA. The IIRC is of the opinion that reliability of information being contained in IRs can be enhanced significantly by systems, such as robust internal control and reporting systems, stakeholder engagement, internal audit and independent external assurance (The International Integrated Reporting Council, 2013). Therefore, in addition to issuing IRs, some firms are also interested in adopting an external assurance to enhance the credibility of the information contained in them. However, it is not clear how external assurance can be conducted. This difficulty arises because there are no widely accepted assurance standards and as it is not clear whether the assurance should be a limited, reasonable or an integrated one. In addition, IRs may take various forms, such as a separately identifiable part of an annual report, a stand-alone report or an expanded management discussion and analysis (Eccles et al., 2019; Maroun, 2017). The motivation behind integrated reporting is transparency and a one-channel communication on firms' performance. IRs may be considered a business card for the firm both externally and internally (Eccles and Saltzman, 2011). It is argued that accounting firms, in particular Big-4, have made significant investment in developing and offering a variety of CSR services. The strong profile of accounting firms as providers of high-quality professional services in the area of corporate reporting is likely to provide greater appeal to stakeholders on their ability to provide highquality assurance services of IRs. Moreover, non-accounting assurance providers are generally small firms, limiting their ability to enjoy scale efficiencies (Hodge *et al.*, 2009). Although non-accounting assurance providers may have better expertise in environmental, social and governance issues, stakeholders are more likely to perceive stronger reliability and credibility in IRs if they are accompanied by assurance statements provided by accounting firms (Simnett et al., 2009b)[8]. In light of the views above, the following hypothesis is proposed:



H2. The positive effect of IRs on financial performance is enhanced by external assurance provided by accounting firms rather than by non-accounting firms.

4. Methodology and econometric model

4.1 Data collection

The study uses two main sources of data:

- (1) financial data were obtained from Compustat North America; and
- (2) data about IRs and external assurance providers came from the GRI website.

Furthermore, a verification of the existence of IRs and external assurance for the GRI sample used in the study was independently conducted. This study's sample of financial and non-financial data includes the periods from 2011 to 2016. There were a total of 106 IRs in GRI data. The practice of IRs is a matter of degree. There are firms that are issuing reports that could be considered IRs, but they do not describe them as integrated. Accordingly, in the review of the firms in the sample, any report made by firms containing financial and non-financial disclosures is considered to be an IR (Eccles and Serafeim, 2011). Furthermore, any form of review provided by an external party is viewed as providing some level of assurance. Overall, 19 IRs received assurance from accounting firms (38 per cent); 31 were assured by non-accounting firms (62 per cent).

4.2 Econometric model and variables

To test the hypotheses, we use the following model. In cross-sectional data sets, heteroscedasticity could be an issue. FGLS is a suggested alternative to OLS when there is evidence of heteroscedasticity (Wooldridge, 2006)[9]. So, we test the model below using both OLS and FGLS estimators:

$$\begin{aligned} \text{FINPERF}_{\text{it}} &= \alpha_0 + \alpha_1 \, \text{IR}_{\text{it}-1} + \alpha_2 \, \text{AP}_{\text{it}-1} + \alpha_3 \, \text{IR}_{\text{it}-1} \, \times \, \text{AP}_{\text{it}-1} + \alpha_4 \text{CONTROLS}_{\text{it}-1} \\ &+ \sum \text{INDUSTRY}_{\text{it}} + \sum \text{YEAR}_{\text{it}} + \varepsilon_{\text{it}} \end{aligned}$$

4.2.1 Dependent variables. Although the findings of previous studies on the relationship of financial performance with CSR disclosures are rather mixed, it would be fairly said that the majority of these studies provide evidence of a positive association between the two (Barnea and Rubin, 2010). In the model, financial performance (FINPERF) is computed by three indicators: stock price growth (GRWSP) (Coram *et al.*, 2009; Edwards and Hilton, 1966), return on equity (ROE) (Artiach *et al.*, 2010; Cormier and Magnan, 2007; Plumlee *et al.*, 2015) and return on assets (ROA) (Artiach *et al.*, 2010; Choi and Wang, 2009; Villalonga, 2004). Although the GRWSP refers to a market-based measure of financial performance, ROE and ROA are accounting-based measures of performance which are used as proxies for support by a broader group of stakeholders.

4.2.2 Independent and control variables. All of the independent variables are one-yearlagged and, with the exception of dummies, are measured in natural logarithms[10]. AP refers to a dummy variable for external assurance (ACC = accounting firm or NONACC = non-accounting firm). IR is a dummy variable which takes on a value of 1 if the firm issues an integrated report. IR \times AP is the interaction of IR and AP. CONTROLS refer to control variables that consist of financial leverage (LEV) (Bhandari, 1988; Dimitrov and Jain, 2008;



Empirical analysis on North American firms

SAMPJ 11,2

Gomes and Schmid, 2010; Modigliani and Miller, 1958), sales revenue (SALES) (Casey and Grenier, 2015; King and Lenox, 2002; Konar and Cohen, 2001; Rajan and Zingales, 1995), firm size (SIZE) (Becchetti *et al.*, 2008; McWilliams and Siegel, 2000; Orlitzky, 2001), capital intensity (CAPIN) (Elmasr, 2007; Servaes and Tamayo, 2013) and (the lack of) capital efficiency (EFF) (Servaes and Tamayo, 2013). INDUSTRY is a dummy variable for different industries in the sample while YEAR is a dummy variable for years 2011-2016[11].

326 5. Results of empirical analyses

5.1 Univariate analysis

Descriptive statistics are reported in Table I. For the entire sample in Table I, the mean of GRWSP is 0.043 with minimum and maximum values of -1.810 and 2.510. SP (stock price) has a mean of \$53.679. Of the two accounting-based measures of financial performance, the mean of ROE is higher than that of ROA (0.205 vs 0.079). IR has a mean of 0.047, suggesting that nearly 5 per cent of firms in the sample have IRs. The mean of NONACC is higher than that of ACC (0.118 vs 0.077), which implies that non-accounting firms issue more assurance on IRs than accounting firms in the sample. Table I presents the descriptive statistics for firms with and without IRs. Firms that issue IRs have higher GRWSP (0.048 vs 0.042), SP (\$58.159 vs \$53.444) and ROE (0.509 vs 0.189) than those without IRs. Table I shows that non-accounting firms (NONACC) provide more assurance on IRs than accounting firms (NONACC) provide more assurance on IRs than accounting firms (ACC) (0.315 vs 0.164).

Table II displays correlations matrix for the entire sample. The correlation coefficients suggest that there is no multicollinearity[12]. While GRWSP is significantly and negatively correlated with ACC ($\rho = -0.0672$), there is no significant correlation of GRWSP with NONACC ($\rho = 0.0117$). Spearman correlations of GRWSP with ACC and NONACC are similar to those of Pearson ($\rho = -0.0758$, 0.0151). IR is positively and significantly correlated with ROE in both Pearson and Spearman correlations of IR with EXAS, ACC and NONACC in both Pearson and Spearman correlations of IR with EXAS, ACC and NONACC in both Pearson and Spearman correlations. The Spearman matrix also shows a positive and significant correlations of IR with LEV and SALES ($\rho = 0.0581$, 0.0945), which suggests that high financial leverage and/or firms with larger sales revenue are more likely to produce IRs.

5.2 Multivariate analysis

As noted previously, cross-sectional data sets might suffer from heteroscedasticity in their error distribution. Therefore, we start the analysis with a test to determine whether the error term has a constant variance. As the test results suggest a heteroscedasticity issue, we use both OLS with robust (heteroscedasticity-corrected) standard errors and FGLS estimators.

Table III presents the estimation results where the growth rate of stock prices (GRWSP) is the dependent variable. IR is significantly and positively related to GRWSP in Regressions (2), (4), (6) and (8), suggesting that integrated reports issued by firms lead to an increase in stock prices by increasing demand for stocks as a result of a reduction in information asymmetry between owners/managers and shareholders (Cosma *et al.*, 2018; Cuadrado-Ballesteros *et al.*, 2017; Frias-Aceituno *et al.*, 2013; Serafeim, 2015). This provides support for *H1*. To put it differently, investors positively value non-financial disclosures released by managements, particularly when they are combined with financial disclosures in a single report (Berthelot *et al.*, 2012). Moreover, we find that any external assurance (EXAS) is significantly and negatively associated with GRWSP (coef. = -0.036, p < 0.05; coef. = -0.023, p < 0.01). While external assurance provided by an accounting firm (ACC) is highly significant and negatively related to GRWSP (coef. = -0.074, p < 0.01; coef. =



Variables	Observations	Mean	SD	Minimum	Maximum	Empirical
Variables	Observations	Wicall	50	winnindini		analysis on
Full Sample						North
GRWSP	1,463	0.043	0.270	-1.810	2.510	American firms
SP	1,463	53.679	44.890	0.023	530.925	1 million can millio
InROE	1,437	0.135	0.345	-4.580	4.268	
ROE	1,463	0.205	2.130	-10.045	70.385	0.05
lnROA	1,462	0.069	0.160	-3.982	0.570	327
ROA	1,463	0.079	0.103	-1.038	0.768	
IR	1,463	0.047	0.212	0	1	
EXAS	1,463	0.198	0.398	0	1	
ACC	1,463	0.077	0.267	0	1	
NONACC	1,463	0.118	0.323	0	1	
lnLEV	1,461	0.203	0.118	0	0.637	
InSALES	1,462	9.135	1.567	0.996	13.089	
InCAPIN	1,461	6.831	1.494	2.743	11.803	
Sample with IR						
GRWSP	73	0.048	0.197	-0.543	0.713	
SP	73	58.159	42.534	1.166	214.4	
lnROE	72	0.289	0.475	-0.377	2.248	
ROE	73	0.509	1.461	-4.007	8.472	
lnROA	73	0.083	0.062	-0.071	0.226	
ROA	73	0.089	0.069	-0.069	0.254	
IR	73	0.753	0.434	0	1	
EXAS	73	0.480	0.503	0	1	
ACC	73	0.164	0.373	0	1	
NONACC	73	0.315	0.468	0	1	
InLEV	73	0.230	0.089	0.033	0.412	
InSALES	73	9.739	1.123	5.098	11.892	
InCAPIN	73	6.685	1.240	4.401	9.672	
Sample without	IR					
GRWSP	1,390	0.042	0.274	-1.810	2.510	
SP	1,390	53.444	45.013	0.023	530.925	
lnROE	1,365	0.127	0.335	-4.580	4.268	
ROE	1.390	0.189	2.159	-10.045	70.385	
lnROA	1.390	0.068	0.163	-3.982	0.570	
ROA	1,390	0.078	0.104	-1.040	0.768	
IR	1,390	0.010	0 100	0	1	
EXAS	1,390	0.183	0.387	Ő	1	
ACC	1,390	0.073	0.260	Ő	1	
NONACC	1,390	0.107	0.309	Ő	1	
InLEV	1,388	0.202	0.119	0	0.637	
InSALFS	1 389	9103	1 580	0 996	13 089	Table I.
InCAPIN	1 388	6 830	1.500	9.743	11 802	Descriptive statistics
IIICAI IIV	1,000	0.009	1.000	2.740	11.005	Descriptive statistics

-0.073, p < 0.01; coef. = -0.078, p < 0.01; coef. = -0.075, p < 0.01) in Regressions (3), (4), (7) and (8), IR × ACC turns out to be highly significant and positive (coef. = 0.109, p < 0.01; coef. = 0.080, p < 0.01; coef. = 0.125, p < 0.01; coef. = 0.069, p < 0.01), confirming *H2*. The negative association of ACC with GRWSP may be due to the high cost of external assurance provided by accounting firms. This view is supported by Briem and Wald (2018) who report that clients do not have their IRs assured because of high assurance costs[13]. The positive coefficient on the interaction suggests that even if external assurance by accounting firms is

SAMPJ 11.2		GRWSP t	lnROE	lnROA	IR	EXAS	ACC	NONACC	LEV	SALES	CAPIN
,-	GRWSP LnROE LnROA	0.1857* 0.2146*	0.2401* 0.2498*	0.2804* 0.6653*	0.0166 0.0605* 0.0214	-0.0459^{*} -0.0369^{*} -0.0608^{*}	-0.0758* -0.0689* -0.0501	0.0151 0.0191 -0.0276	0.0121 0.0832* 0.0370	0.0684* 0.2645* 0.1583*	-0.1406* -0.2999* -0.4832*
328	EXAS ACC NONACC	$\begin{array}{c} 0.0117 \\ -0.0426 \\ -0.0672^{*} \\ 0.0117 \\ \end{array}$	0.0882* 0.0065 -0.0015 0.0159	$\begin{array}{c} 0.0136\\ 0.0061\\ -0.0125\\ 0.0208\\ \end{array}$	0.1768* 0.0618* 0.1606*	0.1780* 0.5912* 0.7323*	0.0452* 0.5870* -0.1011*	0.1748* 0.7322* -0.1056*	0.0581* 0.0024 0.0248 -0.0086	0.0945* 0.0378 -0.0218 0.0694*	-0.0314 0.0903* 0.1037* 0.0197
	LEV SALES CAPIN	0.0231 0.0612* -0.1027*	0.0591* 0.1152* -0.1468*	0.0404 0.2424* -0.1483* -	0.0251 0.0044 -0.0283	-0.0233 0.0509* 0.0849*	-0.0022 -0.0101 0.1087*	-0.0188 0.0749* - 0.0101	-0.0401 0.0070	-0.0822* 0.0026	0.0133 0.0118
Table II. Correlations matrix	Notes: (* above the) statistica diagonal. 1	lly signifi Pearson c	cant at leasorrelation	st at the 1 coefficien	.0 per cent ts are belo	level. Th w the dia	ne Spearma Igonal	n correlat	tion coeffic	cients are

more costly, it might contribute to stock price performance by enhancing the credibility of IRs owing to a reduction in information asymmetry (Frias-Aceituno et al., 2013). It is argued that accounting firms have access to a larger pool of resources to invest in assurance technologies and systems, which could enhance the quality of their services (Martínez-Ferrero and García-Sánchez, 2018). In contrast to the negative impact of ACC on GRWSP, the results show a significant positive association between NONACC and GRWSP in Regression 6 (coef. = 0.016, p < 0.10). However, IR × NONACC is significantly negative in Regressions 6 and 8 (coef. = -0.061, p < 0.05; coef. = -0.068. p < 0.05), confirming the findings of prior research that financial analysts in North America consider assurance from non-accountants less credible, which could adversely affect the stock price growth (Pflugrath et al., 2011). Overall, the results support the view among academics that investors are interested in non-financial information from CSR reports included in IRs more than ever (Eccles and Krzus, 2010). Furthermore, it can be argued that if firms have their CSR report assured by an external assurance provider and integrate it with audited financial statements in one integrated report, their impact on stakeholders including shareholders will likely be positive, leading to an increase in stock prices (Schadewitz and Niskala, 2010).

In examining control variables, we find significant relationships of LEV, SALES and CAPIN with GRWSP in all regressions, which is consistent with prior studies[14]. In addition, the table includes the values of the variance inflation factor (VIF) and Breusch–Pagan/Cook–Weisberg heteroscedasticity test results. In general, there is evidence of collinearity if the mean VIF is greater than unity or if the largest VIF is greater than 10 (Baum, 2006). Although our mean values of VIF indicate slightly collinearity, the maximum VIF values are less than 10[15]. The Breusch–Pagan/Cook–Weisberg test for each regression indicates a significant degree of heteroscedasticity, which justifies the use of FGLS[16].

Table IV shows a replication of the analysis in Table III using an accounting-based measure of FINPERF (ROE) as the dependent variable (Marti *et al.*, 2015). There are numerous studies that examine the effect of accounting-based financial performance on social responsibility (Griffin and Mahon, 1997; Orlitzky *et al.*, 2003). The results of regressions are consistent with those in Table III. While IR is significantly and positively related to FINPERF in Regressions (4) and (5) (coef. = 0.019, p < 0.05; coef. = 0.143, p < 0.10), there is a significant and negative relationship between them in Regressions (1), (2), (7) and (8). In contrast to our findings in Table III, we find EXAS × IR to be positively and highly



Variables	OLS Reg. 1	FGLS Reg. 2	OLS Reg. 3	FGLS Reg. 4	OLS Reg. 5	FGLS Reg. 6	OLS Reg. 7	FGLS Reg. 8
IR EXAS IR ~ FYAS	0.015 (0.028) -0.036** (0.017) -0.036 + 0.017)	0.049^{***} (0.013) -0.023^{***} (0.008) -0.023 (0.029)	0.015 (0.025)	0.033*** (0.012)	0.023 (0.024)	0.041*** (0.011)	0.006 (0.028)	0.048*** (0.018)
III × LXXX3 ACC III × ACC	(0+0.0) 7+0.0	(770.0) 100.0-	-0.074^{***} (0.025) 0.109** (0.048)	-0.073^{***} (0.015) 0.080^{**} (0.031)			-0.078^{***} (0.026) 0.125^{**} (0.050)	$-0.075^{**}(0.015)$ $0.069^{**}(0.035)$
NONACC					0.002 (0.020)	0.016* (0.009)	-0.008 (0.020)	0.006 (0.009)
$IR \times NONACC$					0.002 (0.052)	$-0.061^{**}(0.028)$	0.022 (0.054)	$-0.068^{**}(0.031)$
LEV	-0.015(0.066)	$-0.042^{**}(0.020)$	-0.012(0.065)	-0.047^{**} (0.020)	-0.011(0.065)	-0.039*(0.021)	-0.027 (0.065)	$-0.062^{***}(0.019)$
SALES	$0.012^{**}(0.006)$	$0.014^{***}(0.002)$	$0.011^{**}(0.006)$	$0.013^{***}(0.002)$	$0.011^{**}(0.006)$	$0.012^{***}(0.002)$	0.013*(0.006)	$0.013^{***}(0.002)$
CAPIN	$-0.014^{***}(0.004)$	$-0.012^{***}(0.001)$	-0.014^{***} (0.004)	$-0.011^{***}(0.001)$	$-0.015^{***}(0.004)$	$-0.011^{***}(0.001)$	-0.010^{***} (0.004)	$-0.008^{***}(0.001)$
CONSTANT	0.043(0.064)	0.025 (0.023)	0.038 (0.064)	0.024(0.021)	0.052(0.064)	0.038(0.024)	0.010 (0.068)	0.008 (0.022)
Observations	1,463	1,463	1,463	1,463	1,463	1,463	1,463	1,463
R^2	0.096		0.099		0.094		0.098	
R^2 -adj.	060.0		0.092		0.087		060.0	
Wald test		2012		2492		1498		2627
$\operatorname{Prob} > \chi^2$		0.000		0.000		0.000		0.000
Mean VIF	1.53		1.36		1.45		1.47	
Max. VIF	2.14		1.92		1.93		1.96	
Breusch-Pagan/Cook-Weisberg test	79.82		69.06		87.93		156.18	
$Prob > \chi^2$	0.000		0.000		0.000		0.000	

المنسارات

Dependent variable is the growth of stock price (GRWSP). Regressions include industry- and year-fixed effects. All variables are defined in Appendix 1

Empirical analysis on North American firms

329

Table III.Regression results

SAMPJ 11,2	FGLS Reg. 8	$\begin{array}{c} -0.030^{***} (0.012) \\ -0.034^{****} (0.006) \\ 0.989^{****} (0.228) \end{array}$	$-0.015^{***}(0.004)$ $0.105^{***}(0.026)$	$0.231^{***}(0.005)$ $0.023^{***}(0.001)$	-0.027 * (0.001) -0.235 * (0.010)	$0.240^{***}(0.012)$	L,400		18058	0.000				ions, respectively.
330	OLS Reg. 7	-0.059^{**} (0.028) -0.065^{**} (0.032) 1.006^{***} (0.224)	-0.021 (0.028) 0.240*** (0.090)	0.249**** (0.093) 0.025*** (0.007)	-0.030^{***} (0.006) -0.309^{***} (0.100)	0.276^{***} (0.103)	0.100	0.0916		Ļ	c4.1 1 98	714.58	0.0000	ast squares estimat lix 1
	FGLS Reg. 6	0.027 (0.021)	-0.011^{***} (0.004) 0.047 (0.032)	0.22/*** (0.012) 0.022*** (0.001) 0.000*** (0.001)	-0.029*** $(0.001)-0.242$ *** (0.009)	0.254*** (0.013)	1,400		1651	0.000				asible generalized le re defined in Append
	OLS Reg. 5	0.143* (0.078)	-0.016(0.028) 0.034(0.115)	$0.270^{***}(0.094)$ $0.026^{***}(0.007)$	-0.0306^{***} (0.100)	0.287*** (0.104)	0.065	0.0571			1.42 1.87	711.25	0.0000	least squares and fe fects. All variables a
	FGLS Reg. 4	0.019^{**} (0.009) -0.035^{***} (0.006) 0.928^{***} (0.232)	(0000) *** 1000	0.221^{***} (0.008) 0.022^{***} (0.001) 0.020^{***} (0.001)	-0.028^{***} (0.001) -0.240^{***} (0.009)	0.245*** (0.012)	1,400		2270	0.000				S refer to ordinary y- and year-fixed ef
	OLS Reg. 3	0.036 (0.042) $-0.063^{**} (0.032)$ $0.909^{***} (0.226)$	000 0) 4447 LO 0	0.254^{***} (0.093) 0.025^{***} (0.007) 0.021^{***} (0.006)	-0.306^{***} (0.009)	0.273^{***} (0.103)	0.096	0.0888			1.34 1.87	715.04	0.0000	< 0.1]. OLS and FGI sions include indust
	FGLS Reg. 2	-0.025^{***} (0.008) -0.021^{***} (0.004) 0.124^{**} (0.048)	1000 07 4444 00 0	0.234^{***} (0.007) 0.022^{***} (0.001) 0.008^{***} (0.001)	-0.025^{***} (0.001) -0.241^{***} (0.010)	0.252*** (0.013)	1,400		3980 0.000	0.000				11; ** $p < 0.05$; * $p < utrk (InROE)$. Regress
	OLS Reg. 1	-0.037* $(0.020)-0.039$ * $(0.023)0.393$ **** (0.105)	1000 07 8995000 0	$0.257^{***}(0.094)$ $0.026^{***}(0.007)$	-0.033^{***} (0.006) -0.315^{***} (0.100)	0.288*** (0.103)	0.077	0.0700		c L	0.5.U	712.44	0.0000	ntheses [*** $p < 0.0$ thm of return on equ
			C									an/Cook-Weisberg test		ist standard errors in pare triable is the natural logari
Table IV. Regression results	Variables	IR EXAS IR × EXAS ACC IR × ACC	NONACC IR × NONAC	LEV SALES	EFF	CONSTANT	R^2	R^2 -adj.	Wald test	$Y \operatorname{rob} > \chi^{-}$	May VIF	Breusch–Pag.	$\operatorname{Prob} > \chi^2$	Notes: Robu Dependent va

significantly associated with ROE, suggesting that any external assurance increases the credibility of an IR and therefore positively impacts FINPERF. Similar to our findings in Table III where GRWSP is the dependent variable, the results show a negative relationships of ROE with ACC (coef. = -0.063, p < 0.05; coef. = -0.035, p < 0.01; coef. = -0.065, p < 0.05; coef. = -0.034, p < 0.01). However, in contrast to its association with GRWSP, NONACC is negatively associated with ROE (coef. = -0.011, p < 0.01 coef. = -0.015, p < 0.01; coef. = 0.928, p < 0.01; coef. = 1.006, p < 0.01; coef. = 0.989, p < 0.01 in Regressions (3), (4), (7) and (8) again suggests that external assurance by accounting firms on an integrated report affects financial performance positively. Although the results show no significant influence of IR × NONACC on FINPERF in Regressions (5) and (6), IR × NONACC is found to have a strong positive effect on FINPERF in Regressions (7) and (8) (coef. = 0.240, p < 0.01; coef. = 0.105, p < 0.01; coef. = 0.101, p < 0.01; coef. = 0.105, p < 0.

In general, the findings for regressions where FINPERF is measured by ROA are similar to those of previous studies. In Table V, there is a positive and significant association between IR and FINPERF in Regressions (2), (4) and (6) (coef. = 0.010, p < 0.10; coef. = 0.009, p < 0.01; coef. = 0.015, p < 0.01). Again, IR × ACC is highly significant and positively associated with FINPERF in all regressions (coef. = 0.082, p < 0.01; coef. = 0.056, p < 0.01; coef. = 0.063, p < 0.01). Consistent with estimations from Table IV, NONACC is significantly and negatively associated with FINPERF (coef. = -0.024, p < 0.01; coef. = -0.016, p < 0.01; coef. = -0.026, p < 0.01; coef. = -0.024, p < 0.01; coef. = -0.016, p < 0.01; coef. = -0.026, p < 0.01; coef. = -0.024, p < 0.01; coef. = -0.013, p < 0.01; coef. = -0.026, p < 0.01; coef. = -0.018, p < 0.01. However, IR × NONACC is found to be positively related to FINPERF in Regressions (6), (7) and (8) (coef. = 0.013, p < 0.05; coef. = 0.041, p < 0.05; coef. = 0.021, p < 0.01). The findings suggest that external assurance provided by either accounting or non-accounting firms appears to contribute positively to the relationship between IR and FINPERF where FINPERF is measured by accounting-based indicators (ROE and ROA).

Overall, regression results are significant and support the hypotheses. Specifically, we find that integrated reports have a positive impact on financial performance, and this impact is enhanced when assurance is provided by accounting firms.

5.3 Robustness tests and additional analyses

5.3.1 Robustness tests. We conduct several robustness tests to ensure that our results are not affected by model misspecification or measurement error. For the sake of brevity, they are not reported. First, we re-estimated our model using first-order autoregressive estimator; second, the model has been replicated using return on common stocks as a proxy for FINPERF. The results are consistent with previous ones. Another potential issue is endogeneity that could arise owing to expanded disclosure policies which take into account other variables besides simple costs and benefits (Cuadrado-Ballesteros et al., 2017; Lee and Yeo, 2016). To detect the probable occurrence of endogeneity, we conduct Durbin–Wu– Hausman test^[17]. When the test results suggest that the problem of endogeneity exists, we adopt the GMM approach (Barton and Waymire, 2004). The GMM that yields heteroscedasticity-robust standard errors is widely used in economics and finance studies (Baum, 2006; Baum et al., 2003). The regression results presented in Table VI support those obtained in OLS and FGLS regressions. In two-stage regression where GRWSP is used as the dependent variable, we find that IR has a significant positive effect on GRWSP (coef. = 0.358, p < 0.10). Among the control variables, while STKTRD and GRWSALES have a strong positive effect on GRWSP (coef. = 0.026, p < 0.01; coef. = 0.413, p < 0.01), the impact of GROWTH on GRWSP is negative (coef. = -0.128, p < 0.01). Accordingly, in addition to stock market development, integrated reports and sales are important for the growth of



Empirical analysis on North American firms

FGLS Reg. 8	0.007 (0.006) -0.006 (0.004) 0.063*** (0.011)	-0.021 *** $(0.002)0.021 $ *** $(0.003)0.100 $ *** $(0.004)0.018 $ *** $(0.000)-0.021 $ *** (0.000)	$-0.430^{***} (0.007)$ $0.260^{***} (0.006)$ 1.513		6821	0000		ions, respectively.
OLS Reg. 7	-0.002 (0.013) $-0.017^{**} (0.008)$ $0.092^{***} (0.018)$	-0.020 \cdots (0.010) 0.041 ** (0.017) 0.111 *** (0.037) 0.022 *** (0.007) -0.023 *** (0.002)	-0.505 *** (0.120) 0.272 *** (0.025) 1.513	0.364 0.358		1.44	1.96 323.68	0.000 ast squares estimati lix 1
FGLS Reg. 6	0.015*** (0.003)	$-0.010^{-4.01}$ (0.002) 0.013** (0.006) 0.105*** (0.004) 0.018*** (0.000) -0.021^{***} (0.000)	-0.433 *** (0.007) 0.262 *** (0.006) 1.513		9973 0.000	00000		asible generalized le re defined in Appenc
OLS Reg. 5	0.014 (0.012)	-0.024^{+} (0.009) 0.024 (0.016) 0.113**** (0.037) 0.022**** (0.007) -0.024^{****} (0.002)	-0.504^{***} (0.120) 0.274^{***} (0.025) 1.513	0.362 0.357		1.41	1.89 291.68	0.000 0.000 least squares and fe fects. All variables a
FGLS Reg. 4	0.009**** (0.003) -0.001 (0.004) 0.056**** (0.011)	$\begin{array}{c} 0.114^{***} \left(0.004 \right) \\ 0.020^{***} \left(0.000 \right) \\ -0.022^{***} \left(0.000 \right) \end{array}$	-0.445^{***} (0.007) 0.252^{***} (0.004) 1.513		9825 0.000	00010		S refer to ordinary y- and year-fixed eff
OLS Reg. 3	0.007 (0.009) - 0.014 * (0.008) 0.082**** (0.015)	$\begin{array}{c} 0.113^{****} \left(0.038 \right) \\ 0.022^{****} \left(0.007 \right) \\ -0.024^{****} \left(0.002 \right) \end{array}$	-0.501^{***} (0.119) 0.271^{***} (0.025) 1.513	0.361 0.356		1.33	1.88 284.30	0.000 0.1). OLS and FGL sions include industr
FGLS Reg. 2	$0.010^{*} (0.006)$ $-0.013^{****} (0.002)$ $0.025^{****} (0.008)$	$\begin{array}{c} 0.101^{***} \left(0.004 \right) \\ 0.018^{***} \left(0.000 \right) \\ -0.021^{***} \left(0.000 \right) \end{array}$	-0.432^{***} (0.007) 0.261^{***} (0.006) 1.513		5394	000.0		1]; $**p < 0.05$; $*p < etcs (InROA). Regress$
OLS Reg. 1	-0.003 (0.013) $-0.022^{****} (0.007)$ $0.053^{****} (0.017)$	0.111 ^{***} (0.037) 0.022 ^{****} (0.007) -0.023 ^{****} (0.002)	$-0.505^{***} (0.120)$ $0.272^{***} (0.025)$ 1.513	0.358 0.358		1.49	2.17 323.55	0.000 antheses (*** $p < 0.0$ thm of return on ass
Variables	IR EXAS IR × EXAS ACC IR × ACC	NOVACC IR × NONACC SALES SALES CAPIN	EFF CONSTANT Observations	R ² R ² -adi	Wald test	Mean VIF	Max. VIF Breusch–Pagan/Cook–Weisberø test	Prob > χ^2 Prob > χ^2 Notes: Robust standard errors in pare Dependent variable is the natural logari
	OLS FGLS OLS FGLS OLS FGLS OLS FGLS Variables Reg.1 Reg.2 Reg.3 Reg.4 Reg.5 Reg.7 Reg.8	OLS FGLS OLS FGLS OLS FGLS OLS FGLS OLS FGLS OLS FGLS FGLS<	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $

www

Variables	2SLS GRWSP Coef.	SE	Variables	1st SL SIR Coef.	SE	Empirical analysis on North
IR STKTRD GROWTH GRWSALES IND_DUMMY CONSTANT	0.358^{*} 0.026^{***} -0.128^{***} 0.413^{***} -0.105^{***} -0.634^{***}	$\begin{array}{c} 0.196\\ 0.006\\ 0.039\\ 0.083\\ 0.024\\ 0.198\end{array}$	ACC NONACC STKTRD GROWTH GRWSALES SIZE GDPGRW EMP_LAG LEV_LAG IND_DUMMY CONSTANT	$\begin{array}{c} 0.049^{*} \\ 0.081^{***} \\ -0.004 \\ -0.025 \\ 0.051^{*} \\ -0.000 \\ -0.005 \\ 0.014^{***} \\ 0.012 \\ 0.042^{***} \\ 0.065 \end{array}$	0.028 0.024 0.010 0.024 0.028 0.004 0.025 0.005 0.018 0.008 0.178	American firms 333
Observation Kleibergen-Paap sta Hansen J statistic <i>p</i> - Endogeneity test <i>p</i> - Notes: Robust sta variables, see Appen	utistic p-value value value ndard errors in paren ndix 1	1199 0.000 0.275 0.075 ntheses (***p <	0.01; **p < 0.05;	*p < 0.1). For de	1199 efinitions of	Table VI. GMM Regression Results

stock prices. In the first-stage regression, both ACC and NONACC are found to be positively and significantly associated with IR (coef. = 0.049, p < 0.10; coef. = 0.081, p < 0.01), suggesting that it is highly probable that firms which produce integrated reports have them assured. Note that both GRWSALES and EMP_LAG have a significant and positive impact on IR (coef. = 0.051, p < 0.10; coef. = 0.014, p < 0.01), suggesting that an increase in sales and employment leads firms to issue integrated reports.

The GMM estimator provides three statistics to test the validity of the selected instruments. The Kleibergen-Paap statistics is a test of under-identification for determining if the excluded instruments are correlated with the endogenous variable. The Hansen-J statistic is a test of over-identification. The test results indicate a joint hypothesis that the instruments are relevant is accepted. This means that the instruments are uncorrelated with the error term, and that the excluded instruments are correctly excluded from the model. Finally, the endogeneity test statistic suggests that the null hypothesis that the instrumented variable is exogenous should be rejected in all regressions.

5.3.2 Additional analyses. In this section, we conduct two additional analyses. The first of these is about the relationship between integrated reports and financial performance in a pollution-prone sector. The second examines the impact of the quality of CSR information, which is a major component of IR.

5.3.2.1 Pollution-prone sector. It is argued that financial markets might be anticipated to react differently to firms in pollution-prone sectors (Dhaliwal *et al.*, 2011; Klassen and McLaughlin, 1996; Fonseca, 2010). Klassen and McLaughlin (1996) provide evidence that stock prices of firms in "dirty industries," which have favorable environmental performance, are likely to rise. Similarly, Feldman *et al.* (1997) find that firms that improve their environmental management system and environmental performance increase their stock price by as much as 5 per cent. While firms in pollution-prone sectors do face increased scrutiny around their potential risks, there is also some evidence that they can mitigate some of this perceived risk. Bragdon and Marlin (1972) conclude that for the pulp and paper



industry, there is a strong association between pollution control and good profit record. Thus, it is important for management to produce reports, which emphasize the relationship of financial performance with compliance of environmental regulations, and the firm's efforts in resource conservation and crisis prevention. Therefore, IRs may be of a great interest to stakeholders of firms in pollution-prone sectors, as these help investors evaluate the evidence of a firm's pollution performance[18]. Prior research shows that adopting a more environmentally and socially responsible policy has a significant and favorable impact on firms' perceived riskiness to investors, and accordingly, their cost of capital and value in the market place (Feldman *et al.*, 1997). Moneva and Cuellar (2009) find that investors consider financial rather than non-financial disclosures. Thompson and Cowton (2004) and Goss and Roberts (2011) provide evidence that creditors consider environmental risk in lending decisions. Additionally, firms that are more socially responsible have better conditions to issue equity (Lamont *et al.*, 2001). Besides achieving good performance on environmental issues, it is also important to communicate these achievements to the stakeholders.

IRs provide a more holistic view of business than conventional financial reports by explicitly acknowledging interconnections between environmental, social and financial dimensions of corporate performance (The International Integrated Reporting Council, 2003). The IIRC's integrated reporting framework aims to bring together financial, environmental, social and governance information in a clear, concise, consistent and comparable format (The International Integrated Reporting Council, 2013; de Villiers *et al.*, 2014)). External assurance is important to enhance the credibility of environmental disclosures for firms in pollution-prone sectors (Moroney *et al.*, 2012; Simnett *et al.*, 2009b).

In Table VII, regression results of pollution-prone sectors are presented. Consistent with the results for the full sample in Table III, IR is highly significant and positively associated with GRWSP in FGLS regressions with the exception of Regression (4). While ACC is significantly and negatively related to GRWSP in Regressions (3), (4), (7) and (8) (coef. =-0.111, p < 0.01; coef. = -0.118, p < 0.01; coef. = -0.114, p < 0.01; coef. = -0.119, p < 0.01),[19] the interaction terms, IR \times ACC are highly significant and positive (coef. = 0.220, $p < 10^{-10}$ 0.01; coef. = 0.215, p < 0.01; coef. = 0.221, p < 0.05; coef. = 0.164, p < 0.01, suggesting that assurance services provided by accounting firms on IRs are favorably regarded by investors and that the IRs including external assurance affect stock prices positively. On the other hand, the negative coefficient on IR \times NONACC (coef. = -0.117, p < 0.01; coef. = -0.075, p < 0.01 leads us to think that investors do not value IRs assured by non-accounting firms in pollution-prone sectors positively. We also performed an additional investigation to determine whether firms in the pollution-prone sectors, which produce IRs, are different than those that do not. The comparison of these two groups shows no difference in terms of stock price growth (GRWSP) ($\chi^2 = 1.481$ with 1 df; p = 0.2235). However, when an accounting firm provides assurance on IRs, the difference in GRWSP is highly significant (Kruskall–Wallis H test: $\chi^2 = 5.611$ with 1 df; p = 0.018).

5.3.2.2 Quality of IR. The quality of CSR information can influence the value of an IR. This quality may be determined by the adherence level of the CSR report, which reflects the extent to which the GRI Sustainability Reporting Framework and GRI Standards are applied to reports (Frías-Aceituno *et al.*, 2014; García-Sánchez and Noguera-Gámez, 2017b; Global Reporting Initiative, 2014). The adherence levels indicate an objective quality based on objectively measurable characteristics of CSR reports[20]. In CSR reports prepared in accordance with GRI, there are three adherence levels: A, B and C. A "+" is added to an adherence level if the report is externally assured and that this assurance is publicly available (García-Sánchez and Noguera-Gámez, 2017b; Global Reporting Initiative, 2014). In



11.2

SAMPI

Variables	OLS Reg. 1	FGLS Reg. 2	OLS Reg. 3	FGLS Reg. 4	OLS Reg. 5	FGLS Reg. 6	OLS Reg. 7	FGLS Reg. 8
IR EXAS D. J. DVAS	0.037 (0.042) -0.057* (0.030) 0.042	$0.085^{***}(0.027)$ $-0.058^{***}(0.014)$	0.024 (0.031)	0.016 (0.018)	0.049 (0.037)	0.093*** (0.025)	0.023 (0.042)	0.064** (0.028)
III × EAAS ACC IR × ACC	(1-00-0) (1-00-0)	(000.0) 070.0-	-0.111^{***} (0.036) 0.220^{***} (0.079)	-0.118^{***} (0.017) 0.215^{***} (0.052)			-0.114^{***} (0.037) 0.221^{**} (0.087)	$-0.119^{**}(0.017)$ $0.164^{**}(0.057)$
NONACC					-0.008(0.037)	0.012 (0.016)	-0.020(0.038)	-0.005(0.014)
LEV	0.008 (0.165)	0.092* (0.053)	-0.004(0.163)	0.091* (0.049)	0.030 (0.162)	-0.117 = (0.053) $0.134^{**} (0.053)$	-0.006(0.165)	0.088*(0.052)
SALES	0.030^{***} (0.010)	0.031*** (0.004)	$0.029^{***}(0.010)$	0.034*** (0.003)	0.029^{***} (0.010)	0.030*** (0.004)	0.029^{***} (0.010)	0.033*** (0.004)
- CAPIN	-0.043^{***} (0.013)	$-0.047^{***}(0.005)$	-0.038^{***} (0.013)	-0.043^{***} (0.005)	-0.045^{***} (0.012)	-0.051 *** (0.005)	-0.038^{***} (0.013)	$-0.042^{***}(0.005)$
CONSTANT	-0.172(0.160)	$-0.230^{**}(0.091)$	-0.183(0.158)	$-0.281^{***}(0.087)$	-0.156(0.161)	-0.209^{**} (0.094)	-0.185(0.159)	$-0.277^{***}(0.088)$
Observations	534	534	534	534	534	534	534	534
R^2 -overall	0.112		0.116		0.107		0.116	
R^2 -adj.	0.0930		0.0972		0.0877		0.0942	
Wald test		476		2152		547.1		834.6
$Prob > \chi^2$		0.000		0.000		0.000		0.000
Mean VIF	1.48		1.35		1.41		1.42	
Max. VIF	1.89		1.81		1.83		1.85	
Breusch-Pagan/Cook-Weisberg test	116.79		130.56		122.46		130.63	
$\operatorname{Prob} > \chi^2$	0.000		0.000		0.000		0.000	

المنسارات

Empirical analysis on North American firms

335

Table VII.Pollution-pronesectors' regressionresults

SAMPJ 11,2

336

each year, we classify firms into one of three categories with an ordinal number that takes values between 1 and 3, indicating the adherence level of its CSR reports[21].

Table VIII shows the results of regression analysis that examines the relationship between the quality of CSR reports in terms of adherence level (CSRQUA), IR and one-year (GRWSP), two-year (GRW2SP) and three-year (GRW3SP) growth of stock prices[22]. As noted previously, there is a view among academics that investors pay attention to nonfinancial information from CSR reports included in IRs more than ever (Eccles and Krzus, 2010), which implies that the quality of CSR reports reinforces the impact of integrated reports on investors' decisions. In all regressions, we find that both CSRQUA and IR are significantly and negatively related to GRWSP, GRW2SP and GRW3SP. However, CSRQUA \times IR turns out to be significantly positive, suggesting that the impact of IR on one-, two- and three-year growth rate of stock prices is positively influenced by the quality of CSR reports. To put it differently, an increase in the quality of CSR reports of firms that also issue integrated reports, using this CSR information, has a significant positive impact on stock price growth both in the short and long run[23].

6. Conclusions

There is a growing interest among stakeholders for firms to provide a disclosure which combines financial and socially responsible information in a single report. This disclosure could provide investors with management's perspective on the impact of socially responsible actions on financial performance. Integrated reports are in their infancy, particularly in the USA. Therefore, the benefits of providing an integrated report to stakeholders need further investigation. This study examines the relationship of integrated reports and external assurance on financial performance for North American firms for the period 2011-2016. The results show a significant and positive association between integrated reports and financial performance. Moreover, this association is enhanced when assurance of the integrated report is provided by an accounting firm. The findings from this study support the IIRC's view that integrated reports with assurance are important to stakeholders. A limitation of this study is the small number of North American firms creating integrated reports, which makes it difficult to reach conclusions about the impact of IRs on stakeholders with certainty. However, the results indicate that there are directions for future investigation.

This study has a number of implications. First, as suggested by our results, some decisions need to be made about the content of integrated reports. For the analysis, any report made by a firm which includes both financial and non-financial information was viewed as an integrated report. Therefore, the study includes reports of varying content and quality. The significant relationship between these reports and financial performance indicates a need to set some standards concerning their content. This is consistent with recommendations made by other researchers (García-Sánchez and Noguera-Gámez, 2017b; Pflugrath et al., 2011). A second implication of this study concerns the value of assurance. It has always been implied that financial statements require a particular level of assurance. The results also indicate that assurance of integrated reports has an impact on stakeholders with a particular emphasis on shareholders. While the results suggest that an accounting firm's assurance is more beneficial, regulators may need to consider the benefits other types of assurance providers can bring to IRs. As the results are also consistent in environmentally sensitive industries, a level of assurance of their integrated reports may also be relevant for policymakers. A third implication is related to the society's access to information in integrated reports. While it is clear that there is a cost to firms which produce integrated reports, the benefits to society may outweigh these costs. This may go beyond the



$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccc} {\rm CSRQUA} & -0.066^{94**} \left(0.016 \right) & -0.057^{94**} \left(0.004 \right) \\ {\rm IR} & -0.202^{9**} \left(0.100 \right) & -0.186^{94***} \left(0.055 \right) \\ {\rm CSRQUA\times IR} & 0.109^{9***} \left(0.039 \right) & 0.103^{9***} \left(0.023 \right) \\ {\rm LEV} & 0.047 \left(0.138 \right) & -0.014 \left(0.044 \right) \\ {\rm CROWTH} & -0.288^{94**} \left(0.064 \right) & -0.214^{9***} \left(0.023 \right) \\ {\rm CAPIN} & 0.015^{*} \left(0.009 \right) & 0.116^{9***} \left(0.003 \right) \\ {\rm CAPIN} & 0.011 \left(0.009 \right) & 0.116^{9***} \left(0.003 \right) \\ {\rm CAPIN} & 0.018^{9***} \left(0.05 \right) & 0.014^{9****} \left(0.033 \right) \\ {\rm CAPIN} & 0.011 \left(0.009 \right) & 0.014^{9***} \left(0.003 \right) \\ {\rm CAPIN} & 0.018^{9***} \left(0.05 \right) & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.019^{9***} \left(0.05 \right) & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.019^{9***} \left(0.010 \right) & 0.176^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.190 & 0.114^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.190 & 0.116^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.190 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9****} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9***} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9****} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9****} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9****} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9****} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9*****} \left(0.033 \right) \\ {\rm CAPIN} & 0.000 & 0.014^{9**********} \left(0.000 \right) \\ {\rm CAPIN} & 0.000 & 0.000 \\ {\rm CAPIN} & 0.000 & 0.000$	-0.139**** (0.034) -0.278 (0.218) 0.181*** (0.080) -0.243 (0.269)	-0.123^{***} (0.013)		10200
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{ccccc} {\rm CSRQUA \times IR} & 0.109^{\rm prace} (0.039) & 0.103^{\rm stars} (0.029) \\ {\rm LEV} & 0.047 (0.138) & -0.014 (0.044) \\ {\rm SALES} & 0.047 (0.138) & -0.014^{\rm stars} (0.021) \\ {\rm SALES} & 0.015^{\rm st} (0.009) & 0.016^{\rm stars} (0.003) \\ {\rm CAPIN} & -0.238^{\rm stars} (0.064) & -0.241^{\rm stars} (0.023) \\ {\rm CAPIN} & 0.015^{\rm stars} (0.009) & 0.016^{\rm stars} (0.003) \\ {\rm IND_DUMMY} & -0.184^{\rm stars} (0.052) & -0.173^{\rm stars} (0.023) \\ {\rm CONSTANT} & 0.219^{\rm stars} (0.052) & -0.173^{\rm stars} (0.033) \\ {\rm OBservations} & 496 \\ {\rm A}^2 {\rm overall} & 0.190 & 0.116^{\rm stars} (0.033) \\ {\rm A}^2 {\rm odi} & 0.170 & 0.170 \\ {\rm Wald test} & 0.0170 & 835 \\ {\rm Prob} {\rm Y}^2 & 0.000 \\ \end{array} $	$0.181^{**}(0.080)$ -0.243(0.269)	-0.294^{***} (0.109)	-0.196*** $(0.054)-0.072 (0.577)$	-0.194^{***} (0.012) -0.394^{*} (0.206)
$ \begin{array}{c} {\rm GROWTH} & -0.238^{\rm sev} (0.064) & -0.241^{\rm sev} (0.021) & -0.747^{\rm sev} (0.107) & -0.713^{\rm sev} (0.041) & -1.12^{\rm sev} (0.169) & -0.06 (0.530) & 0.00 \\ {\rm SALES} & 0.016^{\rm s} (0.003) & 0.016^{\rm s} (0.003) & 0.021^{\rm sev} (0.033) & 0.006 (0.303) & 0.000 \\ {\rm SALES} & 0.010 & 0.001^{\rm sev} (0.003) & 0.014^{\rm sev} (0.003) & 0.023^{\rm sev} (0.033) & 0.000 \\ {\rm MDDMMT} & -0.184^{\rm sev} (0.052) & -0.173^{\rm sev} (0.023) & 0.016^{\rm sev} (0.033) & -0.475^{\rm sev} (0.303) & 0.000 \\ {\rm Severations} & 0.010 & 0.176^{\rm sev} (0.033) & 0.748^{\rm sev} (0.033) & 0.456^{\rm sev} (0.033) & 0.031^{\rm sev} (0.031) \\ {\rm Observations} & 496 & 316 & 1.171 & 1.7 \\ {\rm Observations} & 0.170 & 0.278^{\rm sev} (0.213) & 0.456^{\rm sev} (0.213) & 0.456^{\rm sev} (0.213) \\ {\rm Observations} & 0.170 & 0.176^{\rm sev} (0.033) & 0.748^{\rm sev} (0.231) & 0.456^{\rm sev} (0.34) & 1.007^{\rm sev} (0.36) & 0.316 \\ {\rm Observations} & 496 & 3.16 & 1.171 & 1.7 \\ {\rm Observations} & 0.170 & 0.256 & 0.316 & 0.1316 \\ {\rm A}^2 {\rm observations} & 0.223 & 0.000 & 0.256 & 0.273 & 467 \\ {\rm A}^2 {\rm observations} & 1.498 & 0.000 & 0.000 & 0.000 & 0.000 \\ {\rm Mean VIF} & 1.337 & 1.739 & 0.000 & 0.000 & 0.000 \\ {\rm Mean VIF} & 1.337 & 1.739 & 0.000 & 0.000 \\ {\rm Mean VIF} & 1.337 & 1.739 & 0.000 & 0.000 \\ {\rm Mean VIF} & 1.337 & 26.41 & 0.000 & 0.000 \\ {\rm Nobes} {\rm X}^2 {\rm Pob} {\rm >} \chi^2 & 0.000 & 0.000 & 0.000 \\ {\rm Mean VIF} & 1.337 & 2.429 & 6.15 & 0.000 \\ {\rm Nobes} {\rm X}^2 {\rm Pob} {\rm >} \chi^2 & 0.000 & 0.000 \\ {\rm Mean VIF} & 1.337 & 2.429 & 6.33 & 0.000 \\ {\rm Nobes} {\rm X}^2 {\rm Pob} {\rm >} \chi^2 & 0.000 & 0.000 \\ {\rm Modes} {\rm X}^2 {\rm Pob} {\rm >} \chi^2 & 0.000 & 0.000 \\ {\rm Modes} {\rm X}^2 {\rm >} {\rm >} 0.000 & 0.000 & 0.000 \\ {\rm Mean VIF} & {\rm Reusch-PaganCook-Weisberg test & 70.28 & 0.001, {\rm >} \chi^2 {\rm >} 0.000 & 0.000 \\ {\rm Modes} {\rm >} \chi^2 & 0.000 & 0.000 \\ {\rm Modes} {\rm >} \chi^2 {\rm >} 0.000 & 0.000 & 0.000 \\ {\rm Modes} {\rm >} \chi^2 {\rm >} 0.000 & 0.000 & 0.000 \\ {\rm Modes} {\rm >} \chi^2 {\rm >} 0.000 & 0.000 & 0.000 \\ {\rm Modes} {\rm >} \chi^2 {\rm >} 0.000 & 0.000 & 0.000 \\ {\rm Modes} {\rm >} \chi^2 {\rm >} 0$	GROWTH -0.238^{serse} (0.064) -0.241^{serse} (0.035) 0.015^{s} (0.03) 0.015^{serse} (0.03) 0.015^{serse} (0.03) 0.016^{serse} (0.03) 0.016^{serse} (0.03) 0.016^{serse} (0.03) 0.014^{serse} (0.033) 0.011^{serse} (0.033) 0.011^{serse} (0.033) 0.014^{serse} (0.033) 0.016^{serse} (0.033) 0.000^{serse} 0.000^{serse} 0.000^{serse Prob Serverall 0.1770 8.35 0.000 0.000 0.000 0.000		$0.187^{***}(0.044)$ $-0.226^{***}(0.058)$	0.128(0.206) -0.548(0.484)	0.243^{***} (0.075) -0.587^{***} (0.089)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	CAPIN 0.011 (0.09) 0.014*** (0.03) IND_DUMMY -0.14^{4***} (0.05) 0.014^{4***} (0.03) IND_DUMMY -0.18^{4***} (0.05) 0.017^{6***} (0.03) CONSTANT 0.219^{4***} (0.10) 0.77^{6***} (0.03) Observations 496 496 R^2 -adj 0.190 0.190 Wald test 0.170 835 Prob χ^2 0.000	-0.747^{***} (0.107) 0.016 (0.019)	-0.713^{***} (0.041) 0.021^{***} (0.006)	-1.122^{***} (0.198) 0.006 (0.030)	-1.093^{***} (0.069) 0.005 (0.006)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		0.053*** (0.016) -0.448*** (0.085)	$0.046^{***}(0.007) -0.473^{***}(0.038)$	0.093 * * (0.030) - 0.455 * (0.170)	$0.091^{***}(0.006) -0.405^{***}(0.068)$
$ \begin{array}{cccccc} 0.282 & 0.00 & 0.0282 & 0.0316 & 0.0316 & 0.0316 & 0.0316 & 0.0316 & 0.000 & 0.0282 & 0.0316 & 0.0190 & 0.0282 & 0.0273 & 462 & 0.0273 & 462 & 0.020 & 0.0100 & 0.000 & 0$	Observations 450 450 R^2 -odi 0.190 835 Wald test 0.170 835 Prob > χ^2 0.000	$0.748^{***} (0.213)$	0.456^{***} (0.094)	1.007^{***} (0.366)	0.619^{***} (0.097)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	R^2 -adj. 0.170 835 Wald test 835 Prob > χ^2 0.000	0.282	010	0.316	T / T
Wald test 835 149397 465 Prob > χ^2 0.000	Wald test 835 Prob > χ^2 0.000	0.256		0.273	
$ \begin{array}{c} \mathrm{Prob} > \chi^{2} & 0.00 & 0.00 & 0.00 \\ \mathrm{Mem} \mathrm{VIF} & 3.22 & 0.000 & 6.15 & 0.00 \\ \mathrm{Mex} \mathrm{VIF} & 13.37 & 17.39 & 6.15 & 0.00 \\ \mathrm{Breusch-Pagar/Cook-Weisberg test} & 70.28 & 0.00 & 0.00 & 0.00 \\ \mathrm{Prob} > \chi^{2} & 0.000 & 0.00 & 0.00 & 0.000 \\ \mathrm{Notes:} Robust standard errors in parentheses (***p < 0.01; **p < 0.05; *p < 0.1) OLS and FGLS refer to ordinary least squares and feasible generalized least squares estimations, resp Dependent variable is the one-year (GRWSP), two-year (GRW2SP) and three-year (GRWSSP) growth of stock price. Regressions include industry and year fixed effects. All variables are detects. All variable$	$Prob > \chi^2$ 0.000		149397		4635
Max. VI: Max. VI: Breusch-Pagan/Cook-Weisberg test 70.28 2429 56.41 $Prob > \chi^2$ 0.000 0.000 0.000 0.000 0.000 0.000 Notes: Robust standard errors in parentheses (*** $p < 0.05$; * $p < 0.1$). OLS and FGLS refer to ordinary least squares and feasible generalized least squares estimations, resp Dependent variable is the one-year (GRWSP), two-year (GRW2SP) growth of stock price. Regressions include industry and year fixed effects. All variables are de	Mean VIF 329	0.00	0.000	515 A	0.000
Breusch-Pagan/Cook-Weisberg test 70.28 24.29 6.33 $Prob > \chi^2$ 0.000 0.000 0.000 0.000 0.000 0.000 Prob standard errors in parentheses (*** $p < 0.01$; ** $p < 0.05$; * $p < 0.1$). OLS and FGLS refer to ordinary least squares and feasible generalized least squares estimations, resp Dependent variable is the one-year (GRWSP), two-year (GRW2SP), and three-year (GRW3SP) growth of stock price. Regressions include industry and year fixed effects. All variables are de	Max. VIF 13.37	17.39		26.41	
Prob > χ^2 0.000 0.00	Breusch-Pagan/Cook-Weisberg test 70.28	24.29		6.33	
Notes: Robust standard errors in parentheses (*** $p < 0.01$; ** $p < 0.05$; * $p < 0.1$), OLS and FGLS refer to ordinary least squares and feasible generalized least squares estimations, resp Dependent variable is the one-year (GRWSP), two-year (GRWSSP), and three-year (GRW3SP) growth of stock price. Regressions include industry and year fixed effects. All variables are de	$Prob > \chi^2 \qquad \qquad 0.000$	0:000		0.000	
Appendix 1	Notes: Robust standard errors in parentheses (*** $p < 0.01$; ** $p < 0.05$; * $p < 0.1$), OLS	and FGLS refer to ordinary lea	ist squares and feasible ge	neralized least squares estin of year fiyed effects. All ya	mations, respectively. triables are defined in

المنسارات المستشارات

Empirical analysis on North American firms

337

Table VIII. Quality of CSR – egression results SAMPJ 11,2 benefits to shareholders as they make investment decisions. These benefits to society may not be expressly represented in market prices. Integrated reports, which include content describing all of a firm's socially responsible actions, may benefit society in a way not captured in market prices. A future study that investigates the implications of integrated reports in various countries could yield results which suggest ways to improve disclosures to stakeholders.

Notes

- 1. This is a general agency problem, as asymmetric information is a central issue in any form of communication by management (Jensen and Meckling, 1976).
- 2. There are some discussions of whether the IIRC is more focused on investors (Flower, 2015).
- 3. We identified reports which combine financial and non-financial information, but did not have a generally accepted format or name like financial reports (Eccles and Serafeim, 2011).
- The concern that IRs should provide shareholders with value is seen by some as abandoning the original mandate for these reports to be valuable for society (Eccles and Saltzman, 2011; Flower, 2015).
- 5. There is no an agreed-upon definition of long term in the finance literature. A three-year period is considered long-term by Serafeim (2015).
- 6. The results indicate that, in addition to high assurance fees, the perception that the external assurance does not add any value to the report is identified as one factor that causes firms not to seek external assurance (Park and Brorson, 2005).
- 7. CSR reports' complex nature requires opinions of experts in many different areas, ranging from accounting to economic and financial data and from management to elaborate industrial plans. According to Scalet and Kelly (2010), engineering expertise is necessary to evaluate environmental and chemical data. Considering the complexity of the services, even the consultation of the external experts by the accounting firm, who takes full responsibility for final opinions, might not necessarily add credibility to CSR reports and their assurance (Manetti and Toccafondi, 2012)
- 8. There is evidence that accounting firms generally prepare higher-quality assurance statements than non-accounting firms (Briem and Wald, 2018; Romero *et al.*, 2014).
- Generalized least squares (GLSs) estimators are suitable when one or more of the assumptions of homoscedasticity and non-correlation of regression errors fail. Cross-section time-series regressions may have unequal variances. GLS estimation is more efficient than OLS estimation, leading to smaller standard errors, narrower confidence intervals and larger *t*-statistics (Cameron and Trivedi, 2010, p. 153, Becketti, 2013, p. 77).
- 10. Sustainability (non-financial) reports are usually issued with a time lag of three to six months (Serafeim, 2015). Therefore, we used one-year-lagged values of the independent variables in the estimations.
- 11. See Appendix 1 for a detailed description of the variables
- 12. Although some correlations (EXAS, ACC and NONACC) are above 0.50, the variables are not used together in the regressions. Also, it is argued that there is no serious threat of multicollinearity if coefficients of correlations are less than ±0.80 (Farrar and Glauber, 1967; Judge *et al.*, 1985). In addition to coefficients of correlations, VIFs in tables 4 through 9 also indicate that there is no collinearity.
- 13. Prior studies find that the cost of assurance provided by accounting firms is usually higher as opposed to other providers (Simnett *et al.*, 2009).



- 14. While Gomes and Schmid (2010), Bhandari (1988) and Modigliani and Miller (1958) provide evidence that leverage is positively associated with stock returns, Dimitrov and Jain (2008) find a negative association between these two (see Appendix 2).
- A rule of thumb states if the maximum VIF is less than 4, there is no need for further American firms investigation (Baum 2006).
- After initial results indicate heteroscedasticity, we use the robust estimator of OLS to control the heteroscedasticity (Baum 2006).
- 17. Durbin–Wu–Hausman test for endogeneity in IV estimation: $\chi^2 = 47.94$, prob ≥ 0.0000 . GMM is an effective method used by researchers in case of endogeneity that could give rise to errors in variables, omitted variables and simultaneous causality (Baum, 2006).
- Like Bragdon and Marlin (1972), Spicer (1978), also examining pulp and paper firms, find that having better pollution control records has higher profitability, larger size and higher price/ earnings ratio.
- 19. Simnett *et al.* (2009) argue that there is usually an increased cost associated with external assurance provided by members of the accounting profession as opposed to other assurance providers, and as such client-firms choose which assurance provider to employ based on a costbenefit analysis. Accordingly, even though an additional assurance entails higher cost, it will be employed as long as its benefit is considered to be greater than its cost. In particular, stakeholders choosing to support firms operating in pollution-prone sectors which face higher environmental and social risks may see higher value in an IR with assurance provided by an accounting firm.
- 20. There is *n* generally agreed-upon definition of quality; it has been defined differently by different scholars (Reeves and Bednar, 1994). While Abbott (1955) and Feigenbaum (1951) define quality as value, Gilmore (1974) and Levitt (1972) describe it as "meeting and/ or exceeding customers' expectations." "[...] improvement must be related to some 'qualities' or characteristics of products, services, processes, or organizations. Since we are talking of quality ... we make reference to those characteristics that are important for the customers, or the stakeholders, or the company itself" (Conti, 2003, p. 4).
- 21. See Appendix 2.
- 22. See Footnote 5
- Although the maximum VIF values are greater than 4, this results from the interaction terms (see footnote: 14).

References

Abbott, L. (1955), Quality and Competition, Columbia University Press, New York, NY.

- Abeysekera, I. (2008), "Intellectual capital practices of firms and the commodification of labour", Accounting, Auditing and Accountability Journal, Vol. 21 No. 1, pp. 36-48.
- Abeysekera, I. (2013), "A template for integrated reporting", Journal of Intellectual Capital, Vol. 14 No. 2, pp. 227-245.
- ACCA (2013), The Draft Framework for Integrated Reporting, International Integrated Reporting Council, Consultation Draft, London.
- ACCA (2017a), "Creating value benefits to investors", ACCA, London, available at: http:// integratedreporting.org/wp-content/uploads/2017/11/CreatingValue_Benefits_to_InvestorsIIRC. pdf.
- ACCA (2017b), Insights into Integrated Reporting Challenges and Best Practice Responses, The Association of Chartered Certified Accountants, available at: www.accaglobal.com



339

North

Empirical

analysis on

SAMPJ 11,2	ACCA (2018), Insights into Integrated Reporting 2.0: walking the Talk, Informative, ACCA (The Association of Chartered Certified Accountants), London, available at: www.accaglobal.com/content/dam/ACCA_Global/professional-insights/Insights-into-integrated-reporting-2-walking-the-talk/pi-insights-IR-2.0.pdf
240	Adams, C. and Narayanan, V. (2007), "Standardizing sustainability reporting", in Unerman, J., Bebbington, J. and O'Dwyer, B. (Eds), <i>Sustainability Accounting and Accountability</i> , Routledge, London, pp. 70-85.
540	Aguinis, H. and Glavas, A. (2012), "What We know and don't know about corporate social responsibility: a review and research agenda", <i>Journal of Management</i> , Vol. 38 No. 4, pp. 932-968.
	Akins, B. (2018), "Financial reporting quality and uncertainty about credit risk among ratings agencies", <i>The Accounting Review</i> , Vol. 93 No. 4, pp. 1-22.
	Akisik, O. and Gal, G. (2014), "Financial performance and reviews of corporate social responsibility reports", <i>Journal of Management Control</i> , Vol. 25 Nos 3/4, pp. 259-288.
	Akisik, O. and Gal, G. (2017), "The impact of corporate social responsibility and internal controls on stakeholders' view of the firms and financial performance", Sustainability Accounting, Management and Policy Journal, Vol. 8 No. 3, pp. 358-385.
	Alexander, G.J. and Buchholz, R.A. (1978), "Corporate social responsibility and stock market performance", <i>Academy of Management Journal</i> , Vol. 21 No. 3, pp. 479-486.
	Amato, L.H. and Amato, C.H. (2012), "Environmental policy, rankings and stock values", <i>Business Strategy and the Environment</i> , Vol. 21 No. 5, pp. 317-325, doi: 10.1002/bse.742.
	Amir, E. and Lev, B. (1996), "Value-relevance of nonfinancial information: the wireless communications industry", <i>Journal of Accounting and Economics</i> , Vol. 22 Nos 1/3, pp. 3-30.
	Arguelles, M., Balatbat, M. and Green, W. (2017), "Is there an early-mover market value effect for signalling adoption of integrated reporting?", Australasian – Centre for Social and Environmental Accounting Research Conference, CPA Australia.
	Artiach, T., Lee, D., Nelson, D. and Walker, J. (2010), "The determinants of corporate sustainability performance", <i>Accounting and Finance</i> , Vol. 50 No. 1, pp. 31-51.
	Asare, S.K. and Wright, A. (2012), "The effect of type of internal control report on users' confidence in the accompanying financial statement audit report", <i>Contemporary Accounting Research</i> , Vol. 29 No. 1, pp. 152-175.
	Bagnoli, M. and Watts, S.G. (2016), "Voluntary assurance of voluntary CSR disclosure", <i>Journal of Economics and Management Strategy</i> , Vol. 26 No. 1, pp. 205-230.
	Barnea, A. and Rubin, A. (2010), "Corporate social responsibility as a conflict between shareholders", <i>Journal of Business Ethics</i> , Vol. 97 No. 1, pp. 71-86.
	Baron, R. (2014), The Evolution of Corporate Reporting for Integrated Performance, Background Paper for the 30th Round Table on Sustainable Development, OECD, Paris.
	Barth, M.E. and McNichols, M.F. (1994), "Estimation and market valuation of environmental liabilities relating to superfund sites", <i>Journal of Accounting Research</i> , Vol. 32, pp. 177-209.
	Barton, J. and Waymire, G. (2004), "Investor protection under unregulated financial reporting", <i>Journal of Accounting and Economics</i> , Vol. 38 Nos 1/3, pp. 65-116.
	Baum, C.F. (2006), An Introduction to Modern Econometrics Using Stata, Stata Press, College Station, TX.
	Baum, C.F., Schaffer, M. and Stillman, S. (2003), "Instrumental variables and GMM: estimation and testing", <i>The Stata Journal: Promoting Communications on Statistics and Stata</i> , Vol. 3 No. 1, pp. 1-31.
	Bebbington, J., Larrinaga, C. and Moneva, J.M. (2008), "Corporate social reporting and reputation risk management", <i>Accounting, Auditing and Accountability Journal</i> , Vol. 21 No. 3, pp. 337-361.



www.

Becchetti, L., DiGiacomo, S. and Pinnacchio, D. (2008), "Corporate social responsibility and corporate	
performance: evidence from a panel of US listed companies", Applied Economics, Vol. 40 No. 5,	
pp. 541-567.	

- Becketti, S. (2013), Introduction to Time Series Using Stata, 1st ed., Stata Press, TX.
- Benn, S. and Bolton, D. (2011), Key Concepts in Corporate Social Responsibility, Vol. 1, Sage, Los ^F Angeles.
- Berthelot, S., Coulmont, M. and Serret, V. (2012), "Do investors value sustainability reports? A Canadian study", *Corporate Social Responsibility and Environmental Management*, Vol. 19 No. 6, pp. 355-363.
- Bhandari, L.C. (1988), "Debt-equity ratio and expected common stock returns: empirical evidence", The Journal of Finance, Vol. 43 No. 2, pp. 507-528.
- Bragdon, J.H. and Marlin, J.A.T. (1972), "Is pollution profitable? Environmental virtue and reward: must stiffer pollution controls hurt profits?", *Risk Management*, pp. 9-18.
- Brammer, S., Brooks, C. and Pavelin, S. (2006), "Corporate social performance and stock returns: UK evidence from disaggregate measures", *Financial Management*, Vol. 35 No. 3, pp. 97-116.
- Briem, C.R. and Wald, A. (2018), "Implementing third-party assurance in integrated reports", Accounting, Auditing and Accountability Journal, Vol. 31 No. 5, pp. 1461-1485.
- Brown, T.J. and Dacin, P.A. (1997), "The company and the product: corporate associations and consumer product responses", *Journal of Marketing*, Vol. 61 No. 1, pp. 68-84.
- Cahan, S.F., Charl De Villiers, D.C.J., Naiker, V. and van Staden, C.J. (2016), "Are CSR disclosures value relevant? Cross-country evidence", *European Accounting Review*, Vol. 25 No. 3, pp. 576-611.
- Cameron, C.A. and Trivedi, P.K. (2010), *Microeconometrics Using Stata*, Revised Edition, StataCorp LP, College Station, TX.
- Campbell, A. (1997), "Stakeholders: the case in favour", Long Range Planning, Vol. 30 No. 3, pp. 446-449.
- Carroll, A.B. and Shabana, K.M. (2010), "The business case for corporate social responsibility: a review of concepts, research and practice", *International Journal of Management Reviews*, Vol. 12 No. 1, pp. 85-105.
- Casey, R.J. and Grenier, J.H. (2015), "Understanding and contributing to the enigma of corporate social responsibility (CSR) assurance in the United States", *Auditing: A Journal of Practice and Theory*, Vol. 34 No. 1, pp. 97-130.
- Chaidali, P. and Jones, M.J. (2017), "It's a matter of trust: exploring the perceptions of integrated reporting preparers", *Critical Perspectives on Accounting*, Vol. 48, pp. 1-20.
- Cheng, M.M., Green, W.J. and Ko, J.C.W. (2015), "The impact of strategic relevance and assurance of sustainability indicators on investors' decisions", *Auditing: A Journal of Practice and Theory*, Vol. 34 No. 1, pp. 131-162.
- Cheng, M.M., Green, W.J., Conradie, P., Konishi, N. and Romi, A. (2014), "The international integrated reporting framework: key issues and future research opportunities", *Journal of International Financial Management and Accounting*, Vol. 25 No. 1, pp. 90-119.
- Cho, C.H., Michelon, G., Patten, D.M. and Roberts, R.W. (2014), "CSR report assurance in the USA: an empirical investigation of determinants and effects", *Sustainability Accounting, Management* and Policy Journal, Vol. 5 No. 2, pp. 130-148.
- Choi, J. and Wang, H. (2009), "Stakeholder relations and the persistence of corporate financial performance", *Strategic Management Journal*, Vol. 30 No. 8, pp. 895-907.
- Churet, C., Robeco, S. and Eccles, R.G. (2014), "Integrated reporting, quality of management, and financial performance", *Journal of Applied Corporate Finance*, Vol. 26 No. 1, pp. 56-64.
- Cohen, J.R. and Simnett, R. (2015), "CSR and assurance services: a research agenda", Auditing: A Journal of Practice and Theory, Vol. 34 No. 1, pp. 59-74.
- Conti, T. (2003), "A strategic view of organizational stakeholders", in Conti, T., Kondo, Y. and Watson, H.G. (Eds), *Quality into the 21st Century*, ASQ Quality Press, Milwaukee, WI, pp. 1-20.



analysis on North American firms

341

Empirical

SAMPJ 11,2	Coram, P.J., Monroe, G.S. and Woodliff, D.R. (2009), "The value of assurance on voluntary nonfinancial disclosure: an experimental evaluation", <i>Auditing: A Journal of Practice and Theory</i> , Vol. 28 No. 1, pp. 137-151.
	Cormier, D. and Magnan, M. (2007), "The revisited contribution of environmental reporting to investors' valuation of a firm's earnings: an international perspective", <i>Ecological Economics</i> , Vol. 62 Nos 3/4, pp. 613-626.
342	Cormier, D., Magnan, M. and Morard, B. (1993), "The impact of corporate pollution on market valuation: some empirical evidence", <i>Ecological Economics</i> , Vol. 8 No. 2, pp. 135-155.
	Cosma, S., Soana, M.G. and Venturelli, A. (2018), "Does the market reward integrated report quality?", <i>African Journal of Business Management</i> , Vol. 12 No. 4, pp. 78-91.
	Cox, P., Brammer, S. and Millington, A. (2004), "An empirical examination of institutional investor preferences for corporate social performance", <i>Journal of Business Ethics</i> , Vol. 52 No. 1, pp. 27-43.
	Cuadrado-Ballesteros, B., Martínez-Ferrero, J. and García-Sánchez, I.M. (2017), "Mitigating information asymmetry through sustainability assurance: the role of accountants and levels of assurance", <i>International Business Review</i> , Vol. 26 No. 6, pp. 1141-1156.
	Dahlsrud, A. (2008), "How corporate social responsibility is defined: an analysis of 37 definitions", <i>Corporate Social Responsibility and Environmental Management</i> , Vol. 15 No. 1, pp. 1-13.
	Dando, N. and Swift, T. (2003), "Transparency and assurance: minding the credibility gap", <i>Journal of Business Ethics</i> , Vol. 44 No. 2, pp. 195-200.
	de Villiers, C., Rinaldi, L. and Unerman, J. (2014), "Integrated reporting: insights, gaps and an agenda for future research", Accounting, Auditing and Accountability Journal, Vol. 27 No. 7, pp. 1042-1067.
	de Villers, C., Venter, E.R. and Hsiao, PC.K. (2017), "Integrated reporting: background, measurement issues, approaches and an agenda for future research", <i>Accounting and Finance</i> , Vol. 57 No. 4, pp. 937-959.
	De, S. and Sen, P.K. (2002), "Legal liabilities, audit accuracy and the market for audit services", Journal of Business Finance and Accounting, Vol. 29 Nos 3/4, pp. 353-411.
	Deegan, C., Cooper, B.J. and Shelly, M. (2006), "An investigation of the report assurance statements: UK and European evidence", <i>Managerial Auditing Journal</i> , Vol. 21 No. 4, pp. 329-371.
	Demartini, C. and Trucco, S. (2017), <i>Integrated Reporting and Audit Quality: An Empirical Analysis in the European Setting</i> , electronic resource, Springer eBooks.
	Dhaliwal, D.S., Li, O.Z., Tsang, A. and Yang, Y.G. (2011), "Voluntary nonfinancial disclosure and the cost of equity capital: the initiation of corporate social responsibility reporting", <i>The Accounting Review</i> , Vol. 86 No. 1, pp. 59-100.
	Dimitrov, V. and Jain, P.C. (2008), "Value relevance of changes in financial leverage beyond growth in assets and GAAP earnings", <i>Journal of Accounting, Auditing and Finance</i> , Vol. 23 No. 2, pp. 199-222.
	Du Toit, E. (2017), "The readability of integrated reports", Meditari Accountancy Research, Vol. 25 No. 4, pp. 629-653.
	Dumay, J. and Dai, T. (2017), "Integrated thinking as a cultural control?", <i>Meditari Accountancy Research</i> , Vol. 25 No. 4, pp. 574-604.
	Eccles, R.G. and Krzus, M.P. (2010), One Report – Integrated Reporting for a Sustainable Strategy, John Wiley and Sons, Hoboken, NJ.
	Eccles, R.G. and Krzus, M.P. (2014), The Integrated Reporting Movement: Meaning, Momentum, Motives, and Materiality, John Wiley and Sons, New York, NY.
	Eccles, R.G. and Saltzman, D. (2011), "Achieving sustainability through integrated reporting", Stanford Social Innovation Review, Vol. 9 No. 3, pp. 55-61.
	Eccles, R.G. and Serafeim, G. (2011), "The role of the board in accelerating the adoption of integrated reporting", Director Notes (Conference Board), Cambridge, available at: https://corpgov.law.
ستشارات	المنارق در

harvard.edu/2011/12/17/the-role-of-the-board-in-accelerating-the-adoption-of-integrated-reporting/

- Eccles, R.G., Krzus, M.P. and Ribot, S. (2015), "Meaning and momentum in the integrated reporting movement", *Journal of Applied Corporate Finance*, Vol. 27 No. 2, pp. 8-17.
- Eccles, R.G., Michael, P.K. and Liv, A.W. (2019), Integrated Reporting Requires Integrated Assurance, QFINANCE, p. 14, available at: www.financepractitioner.com/auditing-best-practice/integratedreporting-requires-integrated-assurance?full
- Edwards, C.E. and Hilton, J.G. (1966), "A note on the high-low price average as an estimator of annual average stock prices", *The Journal of Finance*, Vol. 21 No. 1, pp. 112-115.
- Elliott, B.W., Jackson, K.E., Peecher, M.E. and White, B.J. (2014), "The unintended effect of corporate social responsibility performance on investors' estimates of fundamental value", *The Accounting Review*, Vol. 89 No. 1, pp. 275-302.
- Elmasr, H. (2007), "Capital intensity and stock returns", *Journal of Investment Strategy*, Vol. 2 No. 1, pp. 61-65.
- EY (2014), Sustainability Reporting The Time Is Now, Accessed 2018, available at: www.ey.com/ Publication/vwLUAssets/EYSustainability-reporting-the-time-is-now/\$FILE/EY-Sustainabilityreporting-the-time-is-now.pdf
- Farrar, D.E. and Glauber, R.R. (1967), "Multicollinearity in regression analysis: a problem revisited", *The Review of Economics and Statistics*, Vol. 49 No. 1, pp. 92-107.
- Feigenbaum, A.V. (1951), Quality Control: Principles, Practice, and Administration, McGraw-Hill, New York, NY.
- Feldman, S.J., Soyka, P.A. and Ameer, P.G. (1997), "Does improving a firm's environmental management system and environmental performance result in a higher stock price?", *The Journal of Investing*, Vol. 6 No. 4, pp. 87-89.
- Ferguson, A. and Pundrich, G. (2015), "Does industry specialist assurance of non-financial information matter to investors?", Auditing: A Journal of Practice and Theory, Vol. 34 No. 2, pp. 121-146.
- Firth, M. and Smith, A. (1992), "Selection of auditor firms by companies in the new issue market", *Applied Economics*, Vol. 24 No. 2, pp. 247-255.
- Flammer, C. (2013), "Corporate social responsibility and shareholder reaction: the environmental awareness of investors", Academy of Management Journal, Vol. 56 No. 3, pp. 758-781.
- Flower, J. (2015), "The international integrated reporting council: a story of failure", *Critical Perspectives* on Accounting, Vol. 27, pp. 1-17.
- Fonseca, A. (2010), "How credible are mining corporations' sustainability reports? A critical analysis of external assurance statements under the requirements of the international council on mining and metals", *Corporate Social Responsibility and Environmental Management*, Vol. 17 No. 6, pp. 355-370.
- Francis, J. and Schipper, K. (1999), "Have financial statements lost their relevance?", Journal of Accounting Research, Vol. 37 No. 2, pp. 319-352.
- Freeman, R.E. (1984), *Strategic Management: A Stakeholder Approach*, Pitman Publishing, San Francisco, CA.
- Frías-Aceituno, J.V., Rodríguez-Ariza, L. and Garcia-Sánchez, I.M. (2013), "The role of the board in the dissemination of integrated corporate social reporting", *Corporate Social Responsibility and Environmental Management*, Vol. 20 No. 4, pp. 219-233.
- Frías-Aceituno, J.V., Rodríguez-Ariza, L. and Garcia-Sánchez, I.M. (2014), "Explanatory factors of integrated sustainability and financial reporting", *Business Strategy and the Environment*, Vol. 23 No. 1, pp. 56-72.
- Friedman, M. (1970), "The social responsibility of business is to increase its profits", The New York Times Magazine, New York, NY.



analysis on North American firms

Empirical

SAMPJ	Friedman, A.I. and Miles, S. (2002), "Developing stakeholder theory", Journal of Management Studies, Vol. 39 No. 1, pp. 1-21.
11,2	Gal, G. (2018), "Are sustainability disclosures fraudulent?", Çalıyurt, K. T. and Said, R. (Eds), Sustainability and Social Responsibility of Accountability Reports, Springer, Singapore, pp. 51-64.
	Gamble, A. and Kelly, G. (2001), "Shareholder value and the stakeholder debate in the UK", <i>Corporate Governance - Governance</i> , Vol. 9 No. 2, pp. 110-117.
344	García-Sánchez, IM. and Noguera-Gámez, L. (2017a), "Integrated information and the cost of capital", International Business Review, Vol. 26 No. 5, pp. 959-975.
	García-Sánchez, IM. and Noguera-Gámez, L. (2017b), "Integrated reporting and stakeholder engagement: the effect on information asymmetry", <i>Corporate Social Responsibility and</i> <i>Environmental Management</i> , Vol. 24 No. 5, pp. 395-413.
	García-Sánchez, IM., Martínez-Ferrero, J. and García-Benau, MA. (2019), "Integrated reporting: the mediating role of the board of directors and investor protection on managerial discretion in munificent environments", <i>Corporate Social Responsibility and Environmental Management</i> , Vol. 26 No. 1, pp. 29-45.
	Gibson, K. (2000), "The moral basis of stakeholder theory", Journal of Business Ethics, Vol. 26 No. 3, pp. 245-257.
	Gilmore, H.L. (1974), "Product conformance cost", Quality Progress, Vol. 7 No. 5, pp. 16-19.
	Global Reporting Initiative (2013), <i>The External Assurance of Sustainability Reporting. GRI Research and Development Series</i> , Global Reporting Initiative, Amsterdam.
	Global Reporting Initiative (2014), "Sustainability disclosure Database - Data legend", GRI, available at: www.globalreporting.org/SiteCollectionDocuments/GRI-Data-Legend-Sustainability-Disclosure- Database-Profiling.pd
	Gomes, J.F. and Schmid, L. (2010), "Levered returns", The Journal of Finance, Vol. 65 No. 2, pp. 467-494.
	Goss, A. and Roberts, G.S. (2011), "The impact of corporate social responsibility on the cost of bank loans", <i>Journal of Banking and Finance</i> , Vol. 35 No. 7, pp. 1794-1810.
	Grewal, J., Hauptmann, C. and Serafeim, G. (2017), "Material sustainability information and stock price informativeness", Working Paper, Harvard Business School, Cambridge.
	Griffin, J.J. and Mahon, J.F. (1997), "The corporate social performance and corporate financial performance debate: twenty-five years of incomparable research", <i>Business and Society</i> , Vol. 36 No. 1, pp. 5-31.
	Griffin, P.A. Lont, D.H. and Sun, Y. (2012), "The relevance to investors of greenhouse gas emissions disclosures", Social Science Research Network. June 20, available at: http://ssrn.com/abstract=1735555
	Herzig, C. and Schaltegger, S. (2006), "Corporate sustainability reporting: an overview", in Martin, B., Roger, B. and Schaltegger, S. (Eds), <i>Sustainability Accounting and Reporting</i> , Dordrecht, London, pp. 301-324.
	Heyes, A.G. (1996), "Lender penalty for environmental damage and the equilibrium cost of capital", <i>Economica</i> , Vol. 63 No. 250, pp. 311-323.
	Hockerts, K. and Moir, L. (2004), "Communicating corporate responsibility to investors: the changing role of the investor relations function", <i>Journal of Business Ethics</i> , Vol. 52 No. 1, pp. 85-98.
	Hodge, K., Subramaniam, N. and Stewart, J. (2009), "Assurance of sustainability reports: impact on report user's confidence and perceptions of information credibility", <i>Australian Accounting Review</i> , Vol. 19 No. 3, pp. 178-194.
	Holm, C. and Birkholm-Laursen, P. (2007), "Risk and control developments in the corporate governance: changing the role of the external auditor", <i>Corporate Governance: An International Review</i> , Vol. 15 No. 2, pp. 322-333.
	Jensen, M.C. and Meckling, W.H. (1976), "Theory of the firm: managerial behavior, agency costs, and ownership structure", <i>Journal of Financial Economics</i> , Vol. 3 No. 4, pp. 305-360.
ستشارات	

www.

- Judge, G.G., Griffiths, W.E., Hill, C.R., Lutkepohl, H. and Lee, T.-C. (1985), *The Theory and Practice of Econometrics*, 2nd ed., Wiley, New York, NY.
- Kaler, J. (2006), "Evaluating stakeholder theory", Journal of Business Ethics, Vol. 69 No. 3, pp. 249-268.
- Kim, I. and Venkatachalam, M. (2011), "Are sin stocks paying the price for accounting sins?", Journal of Accounting, Auditing and Finance, Vol. 26 No. 2, pp. 415-442.
- King, A. and Lenox, M. (2002), "Exploring the locus of profitable pollution reduction", *Management Science*, Vol. 48 No. 2, pp. 289-299.
- Klassen, R.D. and McLaughlin, C.P. (1996), "The impact of environmental management on firm performance", *Management Science*, Vol. 42 No. 8, pp. 1199-1214.
- Knauer, A. and Serafeim, G. (2014), "Attracting long-term investors through integrated thinking and reporting: a clinical study of a biopharmaceutical company", *Journal of Applied Corporate Finance*, Vol. 26 No. 2, pp. 57-64.
- Köbberling, V. and Wakker, P.P. (2004), "A simple tool for qualitatively testing, quantitatively measuring, and normatively justifying savage's subjective expected utility", *Journal of Risk and Uncertainty*, Vol. 28 No. 2, pp. 135-145.
- Konar, S. and Cohen, M.A. (2001), "Does the market value environmental performance?", *Review of Economics and Statistics*, Vol. 83 No. 2, pp. 281-289.
- Lamont, O., Polk, C. and Saá-Requejo, J. (2001), "Financial constraints and stock returns", *Review of Financial Studies*, Vol. 14 No. 2, pp. 529-554.
- Lee, K.-W. and Yeo, G.H.-H. (2016), "The association between integrated reporting and firm valuation", *Review of Quantitative Finance and Accounting*, Vol. 47 No. 4, pp. 1221-1250.
- Levitt, T. (1972), "Product-line approach to service", Harvard Business Review, Vol. 50 No. 3, pp. 41-52.
- McCuaig, B. (2010), "Does an integrated report require an integrated audit?", in Eccles, R.G., Cheng, B. and Saltzman, D. (Eds), *The Landscape of Integrated Reporting. Reflections and Next Steps*, Harvard Business School, Cambridge, pp. 140-143.
- McWilliams, A. and Siegel, D. (2000), "Corporate social responsibility and financial performance: correlation or misspecification?", *Strategic Management Journal*, Vol. 21 No. 5, pp. 603-608.
- McWilliams, A. and Siegel, D. (2001), "Corporate social responsibility: a theory of the firm perspective", *Academy of Management Review*, Vol. 26 No. 1, pp. 117-127.
- McNally, M.-A., Cerbone, D. and Maroun, W. (2017), "Exploring the challenges of preparing an integrated report", *Meditari Accountancy Research*, Vol. 25 No. 4, pp. 481-504.
- Madden, T.J., Roth, M.S. and Dillon, W.R. (2012), "Global product quality and corporate social responsibility perceptions: a cross-national study of halo effects", *Journal of International Marketing*, Vol. 20 No. 1, pp. 42-57.
- Magness, V. (2010), "Environmental disclosure in the mining industry: a signaling paradox", Advances in Environmental Accounting and Management, Vol. 4, pp. 55-81.
- Manetti, G. and Becatti, L. (2009), "Assurance services for sustainability reports: standards and empirical evidence", *Journal of Business Ethics*, Vol. 87 No. S1, pp. 289-298.
- Manetti, G. and Toccafondi, S. (2012), "The role of stakeholders in sustainability reporting assurance", *Journal of Business Ethics*, Vol. 107 No. 3, pp. 363-377.
- Maniora, J. (2017), "Is integrated reporting really the superior mechanism for the integration of ethics into the core business model? An empirical analysis", *Journal of Business Ethics*, Vol. 140 No. 4, pp. 755-786.
- Maroun, W. (2017), "Assuring the integrated report: insights and recommendations from auditors and preparers", *The British Accounting Review*, Vol. 49 No. 3, pp. 329-346.
- Marti, C.P., Rovira-Val, M.R. and Drescher, L.G.J. (2015), "Are firms that contribute to sustainable development better financially?", *Corporate Social Responsibility and Environmental Management*, Vol. 22 No. 5, pp. 305-319, doi: 10.1002/csr.1347.



345

North

Empirical

analysis on

American firms

SAMPJ 11,2	Martínez-Ferrero, J. and García-Sánchez, IM. (2017), "Sustainability assurance and cost of capital: does assurance impact on credibility of corporate social responsibility information?", <i>Business Ethics:</i> A European Review, Vol. 26 No. 3, pp. 223-239.
	Martínez-Ferrero, J. and García-Sánchez, IM. (2018), "The level of sustainability assurance: the effects of Brand reputation and industry specialisation of assurance providers", <i>Journal of Business</i> <i>Ethics</i> , Vol. 150 No. 4, pp. 971-990.
346	Martínez-Ferrero, J., García-Sánchez, IM. and Ruiz-Barbadillo, E. (2018), "The quality of sustainability assurance reports: the expertise and experience of assurance providers as determinants", <i>Business Strategy and the Environment</i> , Vol. 27 No. 8, pp. 1181-1196.
	Mock, T.J., Rao, S.S. and Srivastava, R.P. (2013), "The development of worldwide sustainability reporting assurance", <i>Australian Accounting Review</i> , Vol. 23 No. 4, pp. 280-294, doi:10.1111/ auar.12013.
	Mock, T.J., Strohm, C. and Swartz, K.M. (2007), "An examination of worldwide assured sustainability reporting", <i>Australian Accounting Review</i> , Vol. 17 No. 41, pp. 67-77.
	Modigliani, F. and Miller, M.H. (1958), "The cost of capital, corporate finance and theory of investment", <i>The American Economic Review</i> , Vol. 48 No. 3, pp. 261-297.
	Moir, L. (2001), "What do we mean by corporate social responsibility?", Corporate Governance: The International Journal of Business in Society, Vol. 1 No. 2, pp. 16-22.
	Monaghan, P. (2004), "Put up or shut up", in Henriques, A. and Richardson, J. (Eds), The Triple Bottom Line: Does It All Add U Assessing the Sustainability of Business and CSR, Earthscan, London, pp. 142-154.
	Moneva, J.M. and Cuellar, B. (2009), "The value relevance of financial and non-financial environmental reporting", <i>Environmental and Resource Economics</i> , Vol. 44 No. 3, pp. 441-456.
	Moroney, R., Windsor, C. and Aw, Y.T. (2012), "Evidence of assurance enhancing the quality of voluntary environmental disclosures: an empirical analysis", <i>Accounting and Finance</i> , Vol. 52 No. 3, pp. 903-939.
	Morsing, M., Schults, M. and Nielsen, K.U. (2008), "The 'catch 22' of communicating CSR: findings from a Danish study", <i>Journal of Marketing Communications</i> , Vol. 14 No. 2, pp. 97-111.
	Moskowitz, M.R. (1972), "Choosing socially responsible stocks", Business and Society Review Spring, Vol. 72 No. 1, pp. 71-75.
	Öberseder, M., Schlegelmilch, B.B. and Murphy, P.E. (2013), "CSR practices and consumer perceptions", Journal of Business Research, Vol. 66 No. 10, pp. 1839-1851.
	O'Dwyer, B. and Owen, D.L. (2005), "Assurance statement practice in environmental, social and sustainability reporting: a critical evaluation", <i>The British Accounting Review</i> , Vol. 37 No. 2, pp. 205-229.
	O'Dwyer, B., Owen, D. and Unerman, J. (2011), "Seeking legitimacy for new assurance forms: the case of assurance on sustainability reporting", <i>Accounting, Organizations and Society</i> , Vol. 36 No. 1, pp. 31-52.
	Orlitzky, M. (2001), "Does firm size confound the relationship between corporate social performance and firm financial performance?", <i>Journal of Business Ethics</i> , Vol. 33 No. 2, pp. 167-180.
	Orlitzky, M., Schmidt, F.L. and Rynes, S.L. (2003), "Corporate social and financial performance: a meta- analysis", Organization Studies, Vol. 24 No. 3, pp. 403-441.
	Park, J. and Brorson, T. (2005), "Experiences of and views on third-party assurance of corporate environmental and sustainability reports", <i>Journal of Cleaner Production</i> , Vol. 13 Nos 10/11, pp. 1095-1106.
	Perego, P. (2009), "Causes and consequences of choosing different assurance providers: an international study of sustainability reporting", <i>International Journal of Management</i> , Vol. 26 No. 3, pp. 412-425.



- Perego, P. and Kolk, A. (2012), "Multinationals' accountability on sustainability: the evolution of thirdparty assurance on sustainability reports", *Journal of Business Ethics*, Vol. 110 No. 2, pp. 173-190.
- Peters, G.F. and Romi, A.M. (2015), "The association between sustainability governance characteristics and the assurance of corporate sustainability reports", *Auditing: A Journal of Practice and Theory*, Vol. 34 No. 1, pp. 163-198.
- Petrenko, O.V., Aime, F., Ridge, J. and Hill, A. (2016), "Corporate social responsibility or CEO narcissism? CSR motivations and organizational performance", *Strategic Management Journal*, Vol. 37 No. 2, pp. 262-279.
- Pflugrath, G., Roebuck, P. and Simnett, R. (2011), "Impact of assurance and assurer's professional affiliation on financial analysts' assessment of credibility of corporate social responsibility information", *Auditing: A Journal of Practice and Theory*, Vol. 30 No. 3, pp. 239-254.
- Plumlee, M., Brown, D., Hayes, R.M. and Marshall, S.R. (2015), "Voluntary environmental disclosure quality and firm value: further evidence", *Journal of Accounting and Public Policy*, Vol. 34 No. 4, pp. 336-361.
- Polonsky, M.J. (1995), "A stakeholder theory approach to designing environmental marketing strategy", *Journal of Business and Industrial Marketing*, Vol. 10 No. 3, pp. 29-47.
- Rajan, R.G. and Zingales, L. (1995), "What do We know about capital structure? Some evidence from international data", *The Journal of Finance*, Vol. 50 No. 5, pp. 1421-1460.
- Reeves, C.A. and Bednar, D.A. (1994), "Defining quality: alternatives and implications", Academy of Management Review, Vol. 19 No. 3, pp. 419-445.
- Remgro (2017), 2017 Integrated Annual Report, Integrated annual report, Remgro, p. 155, available at: www.remgro.com
- Romani, S., Grappi, S. and Bagozzi, R. (2013), "Explaining consumer reactions to corporate social responsibility: the role of gratitude and altruistic values", *Journal of Business Ethics*, Vol. 114 No. 2, pp. 193-206.
- Romero, S., Fernandez-Feijoo, B. and Ruiz, S. (2014), "Perceptions of quality of assurance statements for sustainability reporting", *Social Responsibility Journal*, Vol. 10 No. 3, pp. 480-499.
- Scalet, S. and Kelly, T.F. (2010), "CSR rating agencies: what is their global impact?", *Journal of Business Ethics*, Vol. 94 No. 1, pp. 69-88.
- Schadewitz, H. and Niskala, M. (2010), "Communication via responsibility reporting and its effect on firm value in Finland", *Corporate Social Responsibility and Environmental Management*, Vol. 17 No. 2, pp. 96-106.
- Schaub, M. (2006), "Investor overreaction to going concern audit opinion announcements", Applied Financial Economics, Vol. 16 No. 16, pp. 1163-1170.
- Schneider, J., Campbell, D., Vargo, C. and Hall, R. (2011), "An analysis of reported sustainability-related efforts in the petroleum refining industry", *Journal of Corporate Citizenship*, Vol. 44 No. 1, pp. 69-84.
- Scholtens, B. and Zhou, U. (2008), "Stakeholder relations and financial performance", Sustainable Development, Vol. 16 No. 3, pp. 213-232.
- Serafeim, G. (2015), "Integrated reporting and investor clientele", Journal of Applied Corporate Finance, Vol. 27 No. 2, pp. 34-51.
- Servaes, H. and Tamayo, A. (2013), "The impact of corporate social responsibility on firm value: the role of customer awareness", *Management Science*, Vol. 59 No. 5, pp. 1045-1061.
- Sierra-García, L., Zorio-Grima, A. and García-Benau, M.A. (2015), "Stakeholder engagement, corporate social responsibility and integrated reporting: an exploratory study", *Corporate Social Responsibility and Environmental Management*, Vol. 22 No. 5, pp. 286-304.



Empirical analysis on North American firms

SAMPJ	Simnett, R. and Huggins, A.L. (2015), "Integrated reporting and assurance: where can research add value?", <i>Sustainability Accounting, Management and Policy Journal</i> , Vol. 6 No. 1, pp. 29-53.
11,2	Simnett, R., Nugent, M. and Huggins, A.L. (2009a), "Developing an international assurance standard on greenhouse gas statements", <i>Accounting Horizons</i> , Vol. 23 No. 4, pp. 347-363.
	Simnett, R., Vanstraelen, A. and Chua, W.F. (2009b), "Assurance on sustainability reports: an international comparison", <i>The Accounting Review</i> , Vol. 84 No. 3, pp. 937-967.
348	Spicer, B.H. (1978), "Market risk, accounting data and companies' pollution control records", <i>Journal of Business Finance and Accounting</i> , Vol. 5 No. 1, pp. 67-83.
	The International Integrated Reporting Council (2003), <ir> Integrated Reporting, Reporting Framework, The International Integrated Reporting Council, London, pp. 1-35, availabel at: www.theiirc.org</ir>
	The International Integrated Reporting Council (2013), The International <ir> Framework, Reporting Framework, The International Integrated Reporting Council, London, pp. 1-35, available at: www.theiirc.org.</ir>
	Thompson, P. and Cowton, C.J. (2004), "Bringing the environment into bank lending implications for environmental lending", <i>The British Accounting Review</i> , Vol. 36 No. 2, pp. 197-218.
	United States Congress (1934), Securities Exchange Act, United States Congress, Washington, DC.
	Vance, S.C. (1975), "Are socially responsible corporations good investment risks?", Management Review, Vol. 64 No. 8, pp. 18-24.
	Vanhamme, J. and Grobben, B. (2009), "'Too good to be true'!". The effectiveness of CSR history in countering negative publicity", <i>Journal of Business Ethics</i> , Vol. 85 No. S2, pp. 273-283.
	Velte, P. and Stawinoga, M. (2017), "Integrated reporting: the current state of empirical research, limitations and future research implications", <i>Journal of Management Control</i> , Vol. 28 No. 3, pp. 275-320.
	Villalonga, B. (2004), "Intangible resources, Tobin's q, and sustainability of performance differences", <i>Journal of Economic Behavior and Organization</i> , Vol. 54 No. 2, pp. 205-231.
	Wallage, P. (2000), "Assurance on sustainability reporting: an auditor's view on auditing", <i>Auditing: A Journal of Practice and Theory</i> , Vol. 19 No. Supplement, pp. 53-65.
	Whitehouse, L. (2006), "Corporate social responsibility: views from the frontline", <i>Journal of Business Ethics</i> , Vol. 63 No. 3, pp. 279-296.
	Wood, D.J. and Jones, R.E. (1995), "Stakeholder mismatching: a theoretical problem in empirical research on corporate social performance", <i>The International Journal of Organizational Analysis</i> , Vol. 3 No. 3, pp. 229-267.
	Wooldridge, J.M. (2006), <i>Introductory Econometrics – A Modern Approach</i> , 3rd ed., Thomson, South-Western, Mason, OH.



Appendix 1

		analysis on
Variables	Description	North
		American firms
FINPERF	FINPERF, which refers to financial performance, is measured by four different variables: GRWSP = stock price growth based on the average of the annual high stock price (SP) and annual low stock price (SP); ROE = The natural logarithm of return on equity [In (net income/total stockholders' equity); ROA = The natural logarithm of return on assets [In(net income/average assets)]; and RET = The natural logarithm of return on common stocks [In ($P_t - P_{t-1} + DIV_t/p_{t-1}$)]*	349
IR	One-year-lagged dummy variable getting the value of 1 if a firm issues an integrated report and 0 otherwise**	
EXAS	One-year-lagged dummy variable getting the value of 1 if a firm has an integrated report assured by a third party assurance provider (accounting or non-accounting firm**	
ACC	One-year-lagged dummy variable getting the value of 1 if a firm gets assurance on integrated report by an accounting firm and 0 otherwise**	
NONACC	One-year-lagged dummy variable getting the value of 1 if a firm gets assurance on integrated report by a non-accounting firm and 0 otherwise**	
LEV	Natural logarithm of one-year-lagged financial leverage computed by the ratio of total debts to total assets*	
SALES	Natural logarithm of sales revenue*	
GRWSALES	Annual growth of sales revenue	
GROWTH	Natural logarithm of the ratio of book value to one-year-lagged market value of equity (book-to-equity).	
SIZE	Firm size measured by natural logarithm of total assets	
CAPIN	Natural logarithm of one year lagged capital intensity computed as total assets to total employees (*)	
EFF	Natural logarithm of one-year-lagged efficiency computed as cost of goods sold to sales (*)	
CDPCRW	Annual economic growth rate***	
STKTRD	Stocks traded total value (% of GDP)***	
STITLE		Table AI.
Notes: *Auth calculation ba	ors' calculation based on Compustat, **Authors' calculation based on GRI data, ***Authors' sed on WDI (World Bank-World Development Indicators)	Definitions of variables



Empirical

SAMPJ 11,2	Appendix 2	
	Adherence level of CSR reports	Rank
350	Firms that disclose CSR information in accordance with A level of GRI guidelines (i.e. these reports are very comprehensive), and Firms that have an external assurance on their disclosed CSR information in accordance with A level of GRI guidelines (A+). Firms that disclose CSR information in accordance with B level of GRI guidelines (i.e. these reports are complete), and Firms that have an external assurance on their disclosed CSR information in accordance with B level of GRI guidelines (B+).	3 2
Table AII.	 Firms that disclose CSR information in accordance with C level of GRI guidelines (i.e. these reports are very basic), and Firms that have an external assurance on their disclosed CSR information in accordance with C level of GRI guidelines (C+). Notes: In preparing this table, we used the information in the paper by García-Sánchez and N Gámez (2017a, 2017b) 	1 oguera-

Corresponding author

Orhan Akisik can be contacted at: akisik@isenberg.umass.edu

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com



Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

